

Temi di Discussione

(Working Papers)

Central bank refinancing, interbank markets, and the hypothesis of liquidity hoarding: evidence from a euro-area banking system

by Massimiliano Affinito





Temi di discussione

(Working papers)

Central bank refinancing, interbank markets, and the hypothesis of liquidity hoarding: evidence from a euro-area banking system

by Massimiliano Affinito

Number 928 - September 2013

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Massimo Sbracia, Stefano Neri, Luisa Carpinelli, Emanuela Ciapanna, Francesco D'Amuri, Alessandro Notarpietro, Pietro Rizza, Concetta Rondinelli, Tiziano Ropele, Andrea Silvestrini, Giordano Zevi. Editorial Assistants: Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

CENTRAL BANK REFINANCING, INTERBANK MARKETS, AND THE HYPOTHESIS OF LIQUIDITY HOARDING: EVIDENCE FROM A EURO-AREA BANKING SYSTEM

by Massimiliano Affinito*

Abstract

This paper tests the hypothesis of liquidity hoarding in the Italian banking system during the 2007-2011 global financial crisis. According to this hypothesis, in periods of crisis, interbank markets stop working and central banks' interventions are ineffective because banks hoard the liquidity injected rather than channelling it on to other banks and the real economy. The test uses monthly data at banking-group level for all intermediaries operating in Italy between January 1999 and August 2011. This is the first paper to use micro data to analyse the relationship between single banks' positions vis-à-vis the central bank and the interbank market. The results show that the Italian interbank market functioned well even during the crisis, and, contrary to widespread conjecture, the liquidity injected by the Eurosystem was intermediated among banks and towards the real economy. This finding is robust to the use of several estimation methods and data on the different segments of the money market.

JEL Classification: G21, E52, C30. **Keywords**: liquidity, financial crisis, central bank refinancing, interbank market.

Contents

1. Introduction	5
2. Empirical strategy	
3. The data	
4. The determinants of central bank refinancing	
5. The determinants of interbank market positions	
6. Central bank refinancing and customer loans	
7. Robustness checks	
8. Conclusions	
References	
Figures and tables	
-	

^{*} Bank of Italy, Economic Research and International Relations.

1. Introduction¹

The global financial crisis that erupted in the summer of 2007 has reminded everyone of the crucial role played by liquidity markets. This paper joins the academic and policy debate, focusing on central bank refinancing and the interbank market.

The two markets are analysed jointly for three main reasons. First, an adequate amount of liquidity in the economy and its efficient intermediation through the banking system are both crucial for the correct functioning of the financial system, the implementation of monetary policy, the orderliness of the payment system and the financing of the real economy (Allen and Carletti, 2008; Adrian and Shin, 2009; Brunnermeier, 2009). Second, the two wholesale liquidity markets are closely linked. Central bank refinancing is the driver of liquidity; the interbank market is the main market for liquidity exchange. Central bank refinancing can be viewed as the *primary liquidity market* and the interbank market as the *secondary liquidity market*, where the liquidity obtained in the primary market is reallocated. Third, the global financial crisis makes their joint analysis particularly topical. Interbank markets have been characterized by a shortening of maturities, repo runs, and sharp contractions in activity in the unsecured and cross-border segments (e.g. Martin *et al.*, 2010; Angelini *et al.*, 2011; Afonso *et al.* 2011; Gorton and Metrick, 2011; ECB, 2012). Major central banks have resorted repeatedly to extraordinary injections of liquidity.

The literature typically shares the idea that banks' demand for central bank liquidity and interbank markets are to be analysed jointly (e.g. Furfine, 2003; Craig and Fecht, 2007; Bindseil *et al.*, 2009). However, to my knowledge, this is the first paper to analyse the relationship between single banks' positions vis-à-vis the central bank and the interbank market with banking-group level data over an extended period that includes the financial crisis. The aim is to perform the first extensive test of the liquidity hoarding hypothesis using micro data.

The liquidity hoarding hypothesis is very common in the specialized press as well as in the academic literature (e.g. The Economist, 2007; Financial Times, 2008; Edlin and Jaffee, 2009; Heider *et al.*, 2009; Brunetti *et al.*, 2011). It posits that central banks are ineffective in times of crisis for two reasons. First, large liquidity injections increase the excess reserves held by banks, which tend to accumulate liquidity in periods of uncertainty and not to lend to other banks or to real

¹ Email address: massimiliano.affinito@bancaditalia.it. I would like to thank for their comments, without implicating them in responsibility, Filippo Altissimo, Paolo Angelini, Luigi Cannari, Giuseppe Cappelletti, Falko Fecht, Giovanni Guazzarotti, Michele Manna, David Marqués-Ibañez, Stefano Neri, Matteo Piazza, Alberto Franco Pozzolo, Michela Rancan, Alessandro Secchi, and participants at the seminar held at the Bank of Italy (Rome, September 2012); at the ECB-FED Conference "Bank Funding - Markets, instruments and implications for corporate lending and the real economy" (Frankfurt, October 2012); at the ECB Workshop "Excess Liquidity and Money Market Functioning" (Frankfurt, November 2012). The opinions expressed are only mine and do not necessarily reflect those of the Bank of Italy or the Eurosystem.

economy. Second, liquidity injections are not effective in restoring interbank lending because the central bank crowds out the interbank market by becoming counterparty in all liquidity transactions. Actually, the literature on liquidity hoarding is not unanimous on the role of central banks in crises. It suggests two reasons why banks might hoard liquidity and the interbank market might freeze: generally heightened riskiness of the borrower banks (counterparty credit risk); and precautionary accumulation of liquidity by the lending banks (liquidity risk).² However, the literature splits into two conflicting views on the role of central banks. Most of the literature holds that central bank intervention remains warranted even when liquidity hoarding occurs³ but some scholars argue that central banks are ineffective in periods of crisis. This paper tests the liquidity hoarding hypothesis extensively in order to assess these conflicting theses and to verify empirically whether central banks have been ineffective during the crisis or effective in sustaining liquidity among banks and, ultimately, credit to the real economy. The findings have significant policy implications given the need for a better understanding of the markets for liquidity and effectiveness of monetary policy during financial crises.

The paper carries out a composite test. First, since liquidity decisions are taken at very short maturities and the direction of casual nexus is uncertain, the paper investigates both possible casual directions between the two wholesale liquidity markets. Second, since in both cases the interdependence between the two markets remains an issue, the paper constantly controls for their mutual endogeneity by means of instrumental variable (IV) regressions. Third, as the hypothesis refers mainly but not exclusively to the wholesale liquidity markets, the paper examines the connections between them as well as with banks lending to the real economy. Fourth, in order to strengthen the analysis of liquidity redistribution within the domestic market and across borders (e.g., see Schnabl, 2012), the paper studies the different segments of the interbank market (extra-group and intra-group, domestic and non-domestic, bilateral and 'via central counterparties') simultaneously and separately.

The analysis studies the effects of the policy of the Eurosystem on the banking system of a major euro-area country, Italy. The focus on the Italian banking system is useful for three reasons. First, it is a leading euro-area banking system. Second, given Italy's bank-based economy, the interbank and bank credit markets are crucial to the financing of the private sector. Third, supervisory reporting requirements in Italy make a large set of bank-level characteristics available. The sample period spans 152 months from the introduction of the single euro-area monetary policy

² Diamond and Rajan (2005); Wu (2008); Michaud and Upper (2008); Diamond and Rajan (2008); Taylor and Williams (2008, 2009); Schwarz (2009); Heider *et al.* (2009); Gale and Yorulmazer (2010); Acharya and Skeie (2011); De Haan and Van den End (2011); Wolman and Ennis (2011); Berrospide (2012).

³ Flannery (1996); Freixas *et al.* (2000); Acharya *et al.* (2008); McAndrews *et al.* (2008); Ashcraft *et al.* (2009); Allen *et al.* (2009); Keister and McAndrews (2009); Acharya and Merrouche (2010); Freixas *et al.* (2011); Afonso *et al.* (2011).

in January 1999 to the onset of the most acute phase of the euro-area sovereign debt crisis in August 2011. After the summer of 2011, the sovereign debt crisis affected the Italian banking system severely. As sovereign bond yields rose and sovereign ratings deteriorated, wholesale funding dried up, banks' ability to access collateralized lending decreased, and their ability to finance the real economy was seriously undermined (Bank of Italy, 2012; Albertazzi *et al.*, 2012; Bofondi *et al.*, 2013). Excluding the most acute phase of the sovereign debt crisis enables me to focus the analysis on the phase of the crisis in which liquidity hoarding was most likely.

The results clearly and robustly contradict the liquidity hoarding hypothesis. They show that central bank liquidity is not accumulated unproductively but rather channelled to the banking system and the economy. Therefore, they demonstrate that central bank interventions are warranted and effective even in periods of crises. In particular, the results show that in Italy during the 2007-2011 crisis the banks that relied more on central bank refinancing lent more both to banks and to firms and households, and that central bank liquidity injections were not hoarded but rather used to speed up both interbank and customer lending. Banks with copious retail funding lent even more, and did not apply for additional unproductive central bank liquidity. Overall, these results provide a unique picture, confirming for the euro-area findings obtained so far only for the US and only by combining a set of papers (McAndrews *et al.*, 2008; Ashcraft *et al.* 2008; Ashcraft *et al.*, 2009; Christensen *et al.*, 2009; Afonso *et al.*, 2011).⁴

The paper refers to three large fields of the literature. First, the analysis corroborates and complements the prevailing view on liquidity hoarding, namely that central bank intervention is justified even in crises involving the interbank markets. Showing that central bank liquidity is not hoarded unproductively but rather channelled, the analysis demonstrates that central bank interventions are not only justified but also effective.

Second, this paper draws on and contributes to the literature on central bank interventions in the interbank market. The literature suggests three main reasons for intervention. In a normally functioning interbank market, in which banks with a liquidity surplus transfer funds to those with a deficit and illiquid but solvent banks can obtain funding, central banks step in only to fine-tune liquidity conditions and, ultimately, very short-term interest rates (e.g. Selgin, 1993; Freixas *et al.* 1999). When the interbank market becomes dysfunctional because of asymmetric information, so that even solvent banks cannot get credit, central banks must step in to solve a market failure. When liquidity dries up, central banks have two unique abilities: to provide liquidity in sufficient amounts

⁴ McAndrews *et al.* (2008), Ashcraft *et al.* (2009), and Christensen *et al.* (2009) find that the liquidity measures adopted by the Federal Reserve were effective during the 2007-08 financial crisis. Afonso *et al.* (2011) find that liquidity hoarding is an unimportant factor in US interbank loans. Ashcraft *et al.* (2008) show that, during the first phase of the great financial crisis, the Federal Home Loan Bank System (a US government-sponsored liquidity provider alternative to the Fed) provided liquidity to depository institutions, which in turn financed the real economy.

in response to abnormal shocks (Bhattacharya and Gale, 1987; Acharya *et al.* 2008) and to diversify risk across many illiquid banks (Flannery, 1996; Rochet and Vives, 2004). This paper contributes by showing that interventions in the interbank market are effective even during periods of crisis.

Third, this paper is related to the literature on banks' participation in central banks' operations.⁵ It too uses bank-specific characteristics to explain the decision to access central bank credit because banks' heterogeneous business activities and risk profiles generate different liquidity needs, but makes an original contribution in several respects. It analyses the determinants of banks' total borrowing from central banks, whereas the literature typically focuses on the determinants of banks' participation in specific types of central bank operations (which have to do also with banks' strategic behaviour at auctions). It uses monthly observations for a long sample period, whereas the literature typically uses high-frequency data but over a short span of time. It covers all banks operating in Italy, including those that never directly access the central bank's liquidity, thus obtaining complementary inferential information and avoiding sample-selection bias, whereas the literature focuses only on those bidding in at least one auction. It explores the role played by a large set of bank-specific characteristics in determining the demand for central bank liquidity, whereas the literature typically focuses on just a few explanatory factors. It uses aggregate banking-group data, which are better suited to investigating liquidity needs and the decision to access central bank liquidity, whereas the literature utilizes individual data. The paper analyses all interbank transactions, including over-the-counter, and studies the effects of the great financial crisis on banks' demand for liquidity, whereas the literature considers only pre-crisis periods, with the partial exception of Cassola et al. (2011), who analyse the link between willingness to pay in the Eurosystem repo auctions and alternative sources of funding during the summer of 2007. Their main conclusions are consistent with those of this paper: they find that the interbank market did not cease to function properly and show that the euro-area banks that, like Italian banks, relied less on Eurosystem funding before August 2007 appear to have suffered less from the crisis.

Finally, the results show that traditional bilateral domestic interbank market is the main segment used to redistribute the central bank liquidity. This is line with Affinito (2012), who demonstrates that these transactions are favoured by the presence of relationship interbank lending. But the results also indicate that banks redistribute abroad, mainly to other members of their groups.

The rest of the paper is organized as follows. Section 2 describes the methodology. Section 3 presents the data. Sections 4-6 report the results, Section 7 summarizes robustness checks, and Section 8 concludes.

⁵ Peristiani (1998); Breitung and Nautz (2001); Nyborg *et al.* (2002); Furfine (2003); Nautz and Oechssler (2003); Nyborg and Strebulaev (2004); Bruno *et al.* (2005); Linzert *et al.* (2006); Linzert *et al.* (2007); Craig and Fecht (2007); Bindseil *et al.* (2009); Ennis and Weinberg (2009); Fecht *et al.* (2011); Armantier *et al.* (2011).

2. Empirical strategy

The paper conducts an empirical test of the hypothesis of liquidity hoarding in Italy during the crisis period 2007-2011. This means analysing both supposed malfunctioning of the interbank market and ineffectiveness of the central bank policy. I carry out a multi-sided test, which combines four features. First, it examines the mutual interactions between the two wholesale liquidity markets. Second, it studies – simultaneously and separately – five segments of the interbank market and the three positions for each (debt, credit, and net). Third, it explores the determinants of central bank refinancing, and all interbank market segments and positions. Fourth, it analyses the effect of the crisis on all of them. In the following, I detail the four components of my empirical strategy.

(1) My empirical strategy for the *mutual interactions* between the two wholesale liquidity markets explores both possible directions of the casual nexus. Banks' liquidity decisions are typically taken at very short maturities, so it is not trivial to infer *a priori* whether a bank treasurer decides first central bank liquidity demand and then his interbank conduct, or vice-versa. It is likely that both may be the case at different moments depending on very short liquidity needs, surpluses, and opportunities. As a consequence, my test requires a two-way analysis and continuous control for endogeneity.

I start by following the standard literature, which generally estimates banks' demand for central bank liquidity and analyses its determinants (e.g. Peristiani, 1998; Breitung and Nautz, 2001; Nyborg *et al.*, 2002; Furfine, 2003; Linzert *et al.*, 2007; Craig and Fecht, 2007; Bindseil *et al.*, 2009; Armantier *et al.*, 2011; Afonso *et al.*, 2011). That is, I take central bank refinancing as the main dependent variable (on the left-hand side of my equation), and the interbank market as the explanatory variable (on the right-hand side). This estimation answers the general question of the characteristics of the banks that ask for central bank liquidity. And, more specifically, whether they redistribute or hoard liquidity. To explicate, if I find that the banks that apply for central bank liquidity are net interbank lenders, I can conclude that central bank liquidity is likely to be demanded by redistributing banks.

Then, I reverse the experiment, estimating interbank market as the dependent variable, and central bank refinancing as the explanatory variable. This runs counter to the standard literature, but has the merit of explicitly addressing the question of whether central bank refinancing spurs interbank lending.

In both cases, I use IV regressions, which are well suited to joint analysis of the primary and secondary liquidity markets because they allow: (i) handling the endogeneity problem, which exists in both casual directions; and (ii) examining *all* the determinants of *all* liquidity markets at the same

time. I complement the analysis with SUR model estimations, because, while the SUR model does not properly instrument the endogenous variable, it does allow for contemporaneous correlation across the different innovations, and estimation of the mutual effect of the different endogenous variables.

In formal terms, my empirical strategy is represented by a system of equations. In the simplest case, I have two equations:

$$\mathbf{y}_{i,t} = \alpha'_{1} \mathbf{x}_{i,t} + \beta'_{1} \mathbf{K}^{\mathbf{R}}_{i,t-1} + \eta'_{1} \mathbf{b}_{i} + \lambda'_{1} \mathbf{p}_{t} + \varepsilon_{i,t}$$

$$\tag{1}$$

$$\mathbf{x}_{i,t} = \beta'_2 \mathbf{K}^{\mathbf{R}}_{i,t-1} + \eta'_2 \mathbf{b}_i + \lambda'_2 \mathbf{p}_t + \varphi'_2 \mathbf{K}^{\mathbf{I}}_{i,t-1} + \xi_{i,t}$$
(2)

where $y_{i,t}$ is the dependent variable in equation 1 (second stage, in terms of the IV model), and $x_{i,t}$ is the endogenous covariate in equation 1 and the dependent variable in equation 2 (first stage in terms of the IV model), where it is instrumented by the matrix of instruments $K_{i,t-1}^{I}$. As noted above, $y_{i,t}$ may represent central bank refinancing (to bank *i* in month *t*), and $x_{i,t}$ the interbank market position; or vice-versa. Of course, the matrix of instruments $K_{i,t-1}^{I}$ differs between the two versions of the IV estimations. In the SUR estimation, $y_{i,t}$ appears simultaneously as a regressor in the second equation of the system. The matrix of regressors $K_{i,t-1}^{R}$, included in both equations, contains many bank characteristics. α_{1} , β_{1} , η_{1} , λ_{2} , η_{2} , λ_{2} , φ_{2} are vectors of coefficients; α_{1} is the coefficient of interest; $\varepsilon_{i,t}$ and $\xi_{i,t}$ are idiosyncratic errors ~ i.i.d. Bank fixed effects b_{i} and month fixed effects p_{t} are always included in order to control for bank-level unobservable characteristics, such as the extent to which different intermediaries are hit by the financial crisis, and to take into account macroeconomic trends and all unobservable time-varying variables.⁶

(2) As far as the *interbank market segments/positions* are concerned, I split the interbank market into five segments.

(i) Domestic Extra-Group, i.e. the traditional *bilateral* interbank transactions carried out *domestically* among banks not belonging to any banking group or belonging to different banking groups.

(ii) Domestic Intra-Group, i.e. domestic transactions among banks belonging to the same group.

- (iii) Cross-Border Extra-Group.
- (iv) Cross-Border Intra-Group.

⁶ The regressors in the matrixes $K_{i,t-1}^{R}$ and $K_{i,t-1}^{I}$ are lagged to avoid new endogeneity in estimating $x_{i,t}$, and to replicate the publication delay needed for mutual assessment by banks. In order to verify the presence of further endogeneity problems, I also experiment with the variable $x_{i,t-3}$, lagged by a quarter, and accordingly use $K_{i,t-4}^{I}$. See details in Section 7. On a similar use of both lags and (bank and time) fixed effects in a panel IV estimation, see for example Berger and Bouwman (2009).

(v) Central Counterparties (CCPs), i.e. *trilateral* extra-group interbank transactions via domestic central counterparties, in which the ultimate counterparty can be a domestic or a non-domestic bank or another non-domestic central counterparty.

The distinction between Extra-Group and Intra-Group exposures is essential in my analysis because only Extra-Group exposures constitute a real liquidity redistribution through the banking system.⁷ The distinction between Domestic and Cross-Border exposures is used to investigate whether liquidity redistribution occurs and whether it occurs domestically and/or cross-border. The distinction between bilateral and trilateral exposures enables to explore the role played by the new segment of CCPs, which gained greatly in importance during the crisis and is purely neither domestic nor cross-border.

For each segment of the interbank market, I analyse singly the gross borrowing side (Debts), the gross lending side (Credits), and the Net Position (Credits less Debts). The purpose of the Net Position analysis is plain: to see whether the banks that borrow from the central bank are net interbank borrowers or lenders, hence whether central bank liquidity goes to liquidity redistributors or hoarders. To exemplify, when in my system of equations $y_{i,t}$ is central bank refinancing and $x_{i,t}$ is the interbank Net Position, if $\alpha_1 > 0$ this means that the banks asking for central bank liquidity are net interbank lenders, i.e. redistributors. Likewise, when $y_{i,t}$ is the interbank Net Position and $x_{i,t}$ is central bank refinancing, again if $\alpha_1 > 0$, I conclude that central bank liquidity injections prompt liquidity redistribution among banks.

The gross variables Debts and Credits are useful as well, in that for the same Net Position they indicate the extent to which banks are using the interbank market.⁸ And concurrent analysis of them provides a complete picture of liquidity markets, enabling me to estimate the determinants of all interbank positions and to check the stability of the control regressors.

In short, the interbank market is analysed using 13 different variables: 3 positions (Debts, Credits, and Net Position) for 4 segments (Domestic Extra-Group; Cross-Border Extra; Cross-Border Intra; and CCPs) plus 1 position for the Domestic Intra-Group segment.⁹ In this sense, the system of equations 1-2 is only exemplificative of the many specifications I run. For example, when two interbank segments are analysed simultaneously, the system is composed of three equations: the

⁷ To exemplify, if banks paradoxically lent only within their own banking groups, the total interbank market would apparently be working, but actually there would be liquidity hoarding at banking group level.

⁸ To exemplify, let us assume a banking system composed of two banks (*A* and *B*) and two months (t_1 and t_2). During t_1 , *A* and *B* do not exchange their liquidity at all, but during t_2 , *A* lends to and borrows from *B* an amount equal to 100. At the end of both months, each bank's Net Position is zero, but in the first month the interbank market is frozen, whereas in the second it is fully operational (*A* and *B* may have mutually financed their temporary liquidity needs at different times during the month).

⁹ In this segment, Credits and Debts are identical, and Net Position is zero by definition. In this case, I do not estimate the effect of the different positions, but I do retain the Domestic Intra-Group Credits (or Debts) to capture whether the banking groups with greater exchange of internal liquidity also have greater recourse to central bank refinancing and to the other segments.

first equation contains two endogenous regressors, and the matrix of instruments $K_{i,t-1}^{I}$ includes instruments for two segments of the interbank market.¹⁰ In this case, the SUR model is again useful because it allows me to estimate the interactions between different endogenous segments.

(3) As far as the *determinants* are concerned, my empirical strategy allows analysis of bank characteristics in the matrix $K_{i,t-1}^{R}$ as determinants both of central bank refinancing and of all interbank market segments/positions. The inclusion of bank characteristics as explanatory variables is in line with all the literature and serves as a control. Moreover, it provides complementary information for testing the hypothesis of liquidity hoarding. For example, when the central bank refinancing equation is estimated, three covariates are interesting. First, the variable measuring loans to retail customers tests whether the banks taking central bank liquidity are or not intermediating onward to the economy. This variable is so important that in analysing it I reverse the experiment, as in the case of the interbank market, to explore whether central bank refinancing prompts bank loans to retail customers. Second, the variable measuring *retail fundraising* ascertains whether the banks taking central bank liquidity are already liquid, and if they are thus accumulating further liquidity. Third, the banks' health variables (capital, profitability) verify whether sound banks are forced to borrow from the central bank (suggesting a possible malfunctioning of the interbank market).

(4) As far as the *impact of the crisis* is concerned, I split my long sample period into two spans, before and after the onset of the crisis, and then repeat all the estimations of all determinants over the two sub-periods.¹¹ This helps to verify the liquidity hoarding hypothesis because it sheds light on the way in which the determinants of all liquidity markets change over the crisis.

3. The data

I have two kinds of key variables: central bank refinancing, and the set of variables measuring the positions in the different interbank market segments. The source of the data is the Bank of Italy's prudential supervisory reports.

My first key variable - central bank refinancing - is the ratio between the total exposures of each bank towards the central bank in each period (gross or net of amounts re-deposited at the central bank) and total assets. Since the Eurosystem implements its monetary policy operations in a decentralised manner (that is, the ECB coordinates the operations and the national central banks

¹⁰ Alternatively, when tests of endogeneity allow, the system retains two equations, and one of the two interbank market segments is included as exogenous.¹¹ As a check, I also use a difference-in-difference approach. See details in Section 7.

carry out the transactions), and banks having establishments (a head office or branches) in more than one member state may access the Eurosystem liquidity through different NCBs, my dataset on the one hand may exclude the liquidity obtained by an Italian bank through the NCB of another country; but on the other hand it includes the liquidity obtained through the Bank of Italy by, say, a French or a German bank that has a branch in Italy. My variable comprises all kinds of exposures: standing facilities, open market operations, and loans granted through the non-standard measures taken by the Eurosystem during the crisis.¹² The distinction by type of central bank loan is irrelevant for my purposes because I analyse the determinants of the overall demand for central bank liquidity regardless of the substitute role of different instruments.¹³

My second set of key variables measures the three positions (Debts, Credits, and Net Position) in the different segments of the interbank market. The data cover all interbank exposures, including over-the-counter.

All the variables are computed aggregating at banking group or independent bank level monthly bank-by-bank data. The aggregation at group level results from the focus of the paper. First, the only proper way to investigate the decision and determinants of access to central bank liquidity is to refer to groups, insofar as a group comprising various banks may decide to resort to central bank liquidity through one, several or all of them, and in any case these transactions are likely to be decided by the parent bank, to fit into a group-specific scheme, and to be affected by group task-sharing. Second, as is argued in Section 2, the Intra-Group exposures must be removed from the interbank market in order to properly analyse the hypothesis of liquidity hoarding.¹⁴

My sample period runs from January 1999, when the single Euro-area monetary policy was established, to August 2011, as the sovereign debts crisis was growing more acute. The number of time periods (months) is therefore t = 1, 2, ..., 152. To determine the effect of the crisis, I split the

¹² The Eurosystem conducts two standard types of operations: standing facilities and open market operations. Open market operations, the most important, include main refinancing, longer-term refinancing, fine tuning, and structural operations. Since August 2007, the Eurosystem has undertaken several temporary unconventional monetary policy measures. These measures include: (i) extension of the maturity of longer-term refinancing operations; (ii) increase in the amount of liquidity provided through longer-term operations; (iii) a fixed rate, full allotment tender procedure, which allows unlimited access to central bank liquidity for eligible institutions subject to adequate collateral; (iv) extension of the eligible collateral accepted in Eurosystem operations. Eurosystem liquidity may be obtained also by non-euro-area banks. For more details, see Cecioni *et al.* (2011), and Eser *et al.* (2012); ECB (2012).

¹³ There are different areas of the literature that deal with the types of central bank loans: to investigate banks' ability to use specific refinancing options; to ascertain whether stabilization can be achieved by open market operations (Goodfriend and King, 1988; Kaufman, 1991) or lending to individual banks (Flannery, 1996; Goodhart, 1999); to see whether a distinction can be made between monetary-policy and lender-of-last-resort operations (Freixas *et al.*, 1999). For my purposes, these distinctions would be misleading. To exemplify, even if one bank's bidding strategy fails or if the Eurosystem mistakenly injects too little liquidity by market operations, the bank can make up the difference by accessing the standing facilities.

¹⁴ The Bank of Italy collects information on gross bilateral interbank exposures (assets and liabilities of each bank), and the identity of every counterparty. In order to separate the Intra-Group exposures, I used information on the identity of each counterparty and its group of affiliation. For the banks that changed group during my sample period, I traced the current group of affiliation in each *t*. Likewise, I computed at banking group level the other regressors in the matrixes $K_{i,t-I}^{R}$ and $K_{i,t-I}^{I}$.

sample period into two sub-periods: before and after August 2007, the consensus date for the onset of the crisis (although I experiment with alternative dates as a check). In the pre-crisis sample, *T* is equal to 103; in the post-crisis, 49. The total number of observations is around 44,500 in the pre-crisis sample and 16,000 in the post-crisis sample. These numbers reflect: (i) the variation in the number of banking groups and independent banks $i = 1, 2, ..., N_t$ in each *t*; (ii) the removal, in order to round off measurement errors and eliminate outliers, of 5 per cent tail observations for each variable.

Figure 1 shows that loans granted by the Eurosystem through the Bank of Italy increase during the crisis. Figure 2 shows that the share of central bank loans in total assets and the number of banks borrowing from the central bank also increase. With regard to interbank market segments, Figures 3 and 4 show that during the crisis: (i) Domestic Extra-Group interbank market exposures are stable; (ii) Cross-Border Extra-Group interbank exposures decrease; and (iii) exposures via CCPs increase (Cappelletti *et al.*, 2011). Table 1 reports the summary statistics of the key variables. Table 2 shows the correlations. Central bank loans tend to be correlated positively with interbank Debts and negatively with interbank Net Positions. However, there are also non-linear effects, indirectly confirming the need for more sophisticated statistical tools.

Table 3 lists the explanatory variables (again aggregated by banking group), tells how they are calculated, and gives their summary statistics. All regressors are natural logarithms, ratios or dummy variables. All the explanatory variables in the matrix $K_{i,t-1}^{R}$ are again drawn from the Bank of Italy's prudential supervisory reports. The matrix $K_{i,t-1}^{I}$ includes my instruments, which change depending on the variable instrumented. When central bank refinancing is instrumented, I simply use its lagged values, but in some checks I also experiment with interbank segments. When the interbank market positions are instrumented, I use two variables capturing the role of rating agencies, taken from Fitch.¹⁵ The variable Rating is coded so as to take values from 1 to 10, from best to worst, plus 11 to designate unrated banks. The variable Banks without Rating, following Angelini *et al.* (2011), is a dummy that takes the value of 1 for banks with no rating and 0 otherwise.

Two further aspects are worth noting. First, I use quantitative measures of central bank policy and interbank market positions, a self-explanatory choice given that what distinguishes this crisis is the amount of liquidity offered by central banks. Moreover, the attention to quantitative aspects has been increasing in the literature on the interbank market (e.g. Furfine, 2003 and 2009;

¹⁵ Angelini *et al.* (2011) find that Fitch ratings are the most informative in the assessment of banks and financial firms. I use four different kinds of credit scores taken from Fitch, all as monthly averages of daily ratings. My first choice is the overall individual rating; the other three (support, long-term and short-term) are used as controls. In the case of banking groups, I use the rating of the parent company.

King, 2008; Dinger and von Hagen, 2009; Cocco *et al.*, 2009), and this approach permits analysis of all Italian interbank exposures, including over-the-counter exposures for which interest rate data are not available.¹⁶ Second, I use end-of-month stocks for all variables because, apart from information on auctions, which could duplicate the frequency of the auctions themselves, the data are not available on a more frequent basis. All the relevant literature does the same; even when it uses data on single liquidity auctions as a dependent variable, it takes monthly or quarterly or yearly data for regressors. Moreover, as the repeated extraordinary injections of central bank liquidity and the non-standard monetary policy measures demonstrate, the central bank credit supplied during the crisis is intended to meet longer-term funding needs and accordingly has a more stable maturity.

4. The determinants of central bank refinancing

As described in Section 2, to test the hypothesis of liquidity hoarding, I start by following the standard literature, which means estimating the banks' demand for central bank refinancing. The determinants correspond to banks' individual characteristics, and crucially to their interbank positions. This estimation verifies how banks that seek central bank liquidity behave in the interbank market, and in particular whether they hoard or redistribute. The results are reported in Table 4 (split of interbank segments), Table 5 (marginal effects), and Table 6 (sum of interbank segments).

First of all, the problem of endogeneity between central bank refinancing and interbank positions, which potentially concerns all five interbank segments, turns out to be empirically relevant for two segments only: Domestic Extra-Group and CCPs. By contrast, the other three segments (Domestic Intra-, Cross-Border Extra- and Cross-Border Intra-Group) are exogenous, and the results do not change whether or not they are instrumented. The sum of all interbank segments proves to be endogenous.¹⁷ This different endogeneity of the various interbank market segments is a first interesting outcome, and suggests not instrumenting the three exogenous interbank market segments (i.e. placing them in the matrix $K^{R}_{i,t-1}$).

The two endogenous and instrumented segments are also those with the greatest economic impact, so the analysis dwells on them at greater length. In particular, since the reliable results of endogenous regressors are obviously the instrumented ones, in Table 4 specifications (1)-(3) show the relevant IV outcomes of the Domestic Extra-Group positions (while crossing out the results of

¹⁶ From an estimation perspective, all the effects of interest rate developments are captured by the bank and month dummies, which are always included.

¹⁷ More technically, as for Domestic Intra-Group, Cross-Border Extra- and Cross-Border Intra-Group, the Durbin-Wu-Hausman test cannot reject the null hypothesis of no endogeneity. But for Domestic Extra-Group and CCPs the test does reject the null. As for the sum of all interbank segments, the test rejects the null, so the whole interbank market is endogenous.

CCPs). By contrast, specifications (4)-(6) present the relevant IV outcomes of CCPs (and cross out the Domestic Extra-Group).¹⁸ Finally, specifications (7)-(9) present the SUR results, where the two interbank segments are estimated simultaneously along with central bank refinancing, so neither is crossed out.¹⁹ All the results of the other regressors are reliable and consistent across all specifications.²⁰ Regressors across all estimation models and specifications are not always statistically significant, but they do provide clear indications because (i) they never change the statistical significance of their sign, even when tested by a broad range of estimation techniques, specifications and robustness checks; and (ii) the magnitude of the marginal effects (Table 5) furnishes univocal economic interpretations.

Summing up, central bank liquidity is obtained by banks that are net interbank lenders. In particular: (i) central bank liquidity is obtained mainly by banks that redistribute it domestically through the Domestic Extra-Group segment; (ii) banks do not use the CCPs segment to redistribute the liquidity of the central bank, but essentially as an auxiliary funding source; however the redistribution effect of the other segments prevails; (iii) central bank liquidity is also obtained by banks that redistribute it abroad, mainly to banks belonging to the same group; (iv) the domestic internal capital market has negligible effects on resort to central bank liquidity; (v) the banks that access central bank liquidity are those with more loans to the economy and less retail funding. These outcomes contradict the liquidity hoarding hypothesis. Sub-Section 4.1 details the results for the key determinants of central bank refinancing (i.e the interbank market segment positions); Sub-Section 4.2 discusses the results of the other determinants.

4.1 Key explanatory variables

Domestic Extra-Group interbank market segment

Domestic Extra-Group Credits and Net Position are significantly negative before the crisis, but significantly positive after it (Table 4, specifications 2-3 and 8-9). That is, with the crisis, the banks more involved in central bank refinancing are characterized by relatively more interbank Credits and net lending positions. The effect is also economically relevant (Table 5).

¹⁸ With regard to validity and strength of instruments, the results of the standard tests corroborate my choices. As for strength, the *F*-statistic of the reduced form is always sufficiently high, being the same also for the coefficients of the instruments (Table 7). As for validity, the Sargan test is passed, even if actually the greater number of instruments derives from the use of two related variables (Banks without Rating and Rating). In this light, in order to further check the robustness of my instruments, I used $x_{i,t-1}$ as an alternative, and results do hold.

¹⁹ The pairs of variables "Debts and Net Position" and "Credit and Net Position" are never estimated in the same specification because of evident problems of collinearity. On the other hand, the two variables Debts and Credits can be included in the same specification. In this case, in order not to weaken my instruments, I employed again $x_{i,t-1}$ as an additional instrument in the matrix $K_{i,t-1}^1$. Results are equivalent and unreported.

²⁰ In all my estimations, the observations are clustered at banking group level (and at bank level for independent banks), thus obtaining heteroskedasticity-robust standard errors and controlling for possible autocorrelations across the same banking group.

CCPs interbank market segment

The opposite effect is observed for CCPs. During the crisis, the banks with more CCPs Credits and those that are net lenders in the segment resort less to central bank liquidity (Table 4, specifications 4-9). However, the positive redistribution effect of the Domestic Extra-Group segment prevails, both in quantitative terms, measured by the marginal effects (Table 5), and when the figures of the two segments are added up and a single IV regression is run instrumenting their sum (Table 6, specifications 10-12).

Domestic Intra-Group interbank market segment

A larger Domestic Intra-Group liquidity market means less recourse to central bank refinancing, both before and after the crisis (Table 4). However, the marginal effect is negligible (Table 5).

Cross-Border Extra-Group and Intra-Group interbank market segments

Banks with more Cross-Border Extra-Group Debts and Credits borrow less from the central bank (Table 4). As for Net Position, the effect changes with the onset of the crisis: after it, Cross-Border Extra-Group interbank net-lenders have greater recourse to central bank refinancing. Even more, banks borrow from the central bank when lending to foreign banks belonging to the same group. These outcomes confirm the cross-border redistribution of Eurosystem liquidity. Since this is particularly true for Cross-Border *Intra-Group* Credits, it confirms that international banking groups raise funds in a decentralised manner (Freixas and Holthausen, 2005; ECB, 2011). However, the marginal effect of these variables is modest (Table 5).²¹

Total secondary liquidity market

Since some interbank segments present mixed results, I also estimated their combined effect to double-check the overall outcome, adding up the figures of four segments, excluding Domestic Intra-Group, and instrumenting this sum. This was done in two steps: first, I added all the variables measuring the external exposures (Domestic Extra-Group, CCPs, and Cross-Border Extra-Group variables), and then also the Cross-Border Intra-Group variables. Again, in both cases, the redistribution effect found in the Domestic Extra-Group segment drives all the others (Table 6, specifications 13-15).

4.2 The other determinants

²¹ The presence of foreign banks impacts on all the variables of my estimations, but it is more likely to matter for the covariates capturing the non-domestic transactions. However, the presence of foreign banks is taken into account through the inclusion of bank fixed effects. Moreover, I run on the issue several robustness checks detailed in Section 7.

Loans

My results signal that the banks that get resources from the central bank are those with a higher incidence of loans not only to other banks but also to the economy. The variable Loans is constantly positive after the crisis (Tables 4-6). This positive effect may be explained in part by their use as collateral in central bank operations. However, while this use is minor as a matter of stylized fact (Bank of Italy, 2011b), the positive estimated economic effect is considerable: in the crisis, climbing from the 25th to the 75th percentile, the variable Loans produces the greatest percentage-point increase in the central bank loan share of total assets (Table 5).²²

Fundraising

The variable Fundraising is always negative, and has a large economic impact (Tables 4-6). Banks with large-scale deposits and retail bond issues have less need for central bank liquidity, even in the crisis, and thus do not accumulate further liquidity.

ROE and Capital

According to Afonso *et al.* (2011), since banks only resort to the central bank if other forms of funding are not accessible, one can argue that if banks with good past performance are forced to borrow from the central bank, this is an alarming sign of dysfunction in the interbank market. My results show that this is not the case. The variable ROE is statistically insignificant in both the preand post-crisis periods (as in Cassola *et al.*, 2011); the variable Capital is always negative. That is, healthy banks are not forced to turn to the central bank refinancing, the same result found by Afonso *et al.* (2011) for the US.

Bad Loans

The variable Bad Loans tends to be negative before the crisis (as in Fecht *et al.*, 2011) and positive in the post-crisis period. This is the only result that supports the liquidity hoarding prediction that the liquidity requirement mainly affects the banks that perform worse (Allen *et al.*, 2009; Acharya *et al.*, 2009; Heider *et al.*, 2009; Acharya and Merrouche, 2010; Acharya and Skeie, 2011). It could also signal a risk of moral hazard and/or a risk-taking channel effect (Adrian and Shin, 2009; Borio and Zhu, 2008). In any case, over my sample period the economic impact of Bad Loans on central bank refinancing is modest (Table 5). Gilbert (1995) and Stojanovic *et al.* (2008) also find a statistically significant yet economically negligible effect of refinancing on banks' risk-

²² In any case, even if the positive effect of Loans were partially due to their use as collateral, my results would still indicate a virtuous circle between central banks' liquidity provisions and Loans, and in any case absence of liquidity hoarding.

taking. Therefore, my results counsel a simple early warning to avert the creation of perverse incentives during phases of massive liquidity injection.

Size

The variable Size tends to be negative before and positive after the crisis. This confirms that in the pre-crisis period the larger banks get funding more easily (Kashyap *et al.* 2002), and are less dependent on participation in central bank auctions (Linzert *et al.*, 2006; Bindseil *et al.*, 2009). By contrast, in the post-crisis period, the larger banks are more severely affected by the restrictive conditions in funding markets and have a greater recourse to central bank refinancing (Ashcraft *et al.*, 2008; Fecht *et al.*, 2011; Bank of Italy, 2011a).

Securities holdings and Securitized Loans

Borrowing from central banks is typically collateralized. However, the Eurosystem accepts a broad range of assets as collateral, and during the crisis it extended the range, so collateral is unlikely to be a limiting factor. In any case, it is interesting to see which of the eligible assets are most commonly posted. The variable Portfolio of Government Debt Securities tends to be positive before the crisis and negative after; that is, the use of government bonds as collateral decreases in the crisis, in part simply because the Eurosystem extended eligibility to other securities (typically, in operations with the central bank, "bad collateral drives out good"; see Ewerhart and Tapking, 2008; ECB, 2012). Conversely, the variables Portfolio of Bank Bonds and Securitized Loans tend to be negative before and positive after the crisis.

5. The determinants of interbank market positions

So far, I have explored the determinants of central bank refinancing and shown that the banks that apply for it are not those that accumulate but those that redistribute their liquidity surpluses. Notably, this is found in a panel context, so the redistribution effect concerns the entire period 2007-2011. Nevertheless, as explained in Section 2, one could still argue that the liquidity hoarding hypothesis must be subjected to a reverse-causation test with central bank refinancing as the determinant/driver of interbank positions.

Such a test can be carried out in the context of the previous exercise exploiting the proprieties of the SUR model. Indeed, as is clarified in Section 2, once suitably specified, the SUR model allows simultaneous estimation of central bank refinancing both as dependent variable and as regressor (i.e. the SUR specifications can include both variables $y_{i,t}$ and $x_{i,t}$ in both equations 1 and 2). Table 7 reports the results for equation 2, which couple with those of equation 1 reported in Table 4. The specifications correspond: specifications (1)-(3) are the first stage IV results of the

Domestic Extra-Group positions; specifications (4)-(6) are the first stage IV results of the CCPs positions. Specifications (7)-(9) are the SUR results of a system of three equations, and refer to both Domestic Extra-Group and CCPs positions. Here the inverted relations show that during the crisis central bank liquidity affects the Domestic Extra-Group Credits and Net Position positively. Again, the outcomes are the opposite of those for CCPs, which however (again) have a much smaller marginal effect (Table 9).

However, the SUR model does not permit instrumentation, so I also run a reversed IV experiment, instrumenting central bank refinancing in the first stage by its lagged values, and then using it as the key explanatory variable to estimate the interbank positions in the second stage.²³ The results are reported in Table 8. During the crisis, central bank refinancing spurs interbank lending significantly, both for the Domestic Extra-Group and the sum of interbank segments, both for gross Credits and Net Position. The economic impact is also notable (Table 9).

I also conduct a new test. So far, I have taken the interbank positions as ratios to total assets, for two reasons: first, in analogy with central bank refinancing, which is normalized by total assets; and second because, given the panel context, the ratios capture at least in part the development. However, as a further check, in Table 10 the interbank positions are again used as dependent variables, but measured as growth rates. The results are substantially equivalent. Some minor changes involve a few control regressors, and are explained by the different measure of the dependent variable. Most important, liquidity injections are found to speed up interbank lending. These findings directly rebut the liquidity hoarding hypothesis, widely held in the crisis; instead they demonstrate the effectiveness of monetary policy in Italy in the crisis years 2007-2011.

In addition to central bank liquidity, my estimations also show the other determinants of interbank market positions (Tables 7-10). Six main findings emerge. (i) The sign of the most of the determinants does not change with the crisis, another outcome that contradicts the thesis of a malfunctioning interbank market. (ii) When the figures for the various segments are added together, the determinants of the sum substantially replicate the determinants of the Domestic Extra-Group segment, which therefore again prevails. (iii) Banks that are net lenders externally are net borrowers domestically. (iv) The effect of a larger Domestic Intra-Group segment on presence in other segments is negligible (as was found in the estimation of central bank refinancing). (v) The relationship between the traditional bilateral Domestic Extra-Group and the trilateral CCPs segments tends to be positive. (vi) The determinants of the positions in the two interbank market

²³ Even in the reversed experiment, the standard statistical tests reject the null hypothesis of no endogeneity and suggest that my instruments are valid and not weak. However, as a further check, I also used the other interbank market segments as instruments for central bank refinancing, and the results do still hold.

segments do not always coincide, which explains why the mutual relationship is positive but the impact on central bank refinancing conflicts. The rest of this section deals with this issue.

The effect is common for four kinds of determinants. (a) The results for Size confirm that larger banks have greater liquidity needs in the crisis: both bilateral and trilateral Debts (Net Position) are increasing (decreasing) in Size. The economic effect is substantial (Table 9). (b) The variables Rating and Banks without Rating (which are to be considered together) corroborate the hypothesis of peer monitoring (e.g. Furfine, 2001; King, 2008), as lower-rated banks receive less funds, both bilateral and trilateral. (c) The negative effect of Cross-Border (Extra- and Intra-Group) Net Position on both Domestic Extra-Group and CCPs Net Position confirms that banks that are net lenders externally (in particular Intra-Group) are net borrowers domestically. (d) As to Credits only, Capital and Fundraising have an identical effect in the two segments of the interbank market. Highly capitalized banks lend less in both segments, probably because they have greater capability for locating profitable investment opportunities outside the interbank market. Banks with more funds from their retail customers lend more in both segments, another result indicating that the more liquid banks do not hoard.

For three determinants the effect is different. (a) The effect of Fundraising is positive for Domestic Extra-Group Net Position (i.e. the more liquid banks are net interbank lenders); but it is negative, though smaller, for CCPs. This confirms that the Domestic Extra-Group segment, but not the CCPs, is used to redistribute liquidity among banks. (b) Banks with more Loans (to customers) conceivably borrow more (and lend less) in the traditional bilateral interbank segment, but they borrow less via CCPs and are net lenders in this interbank segment. (c) The variable Bad Loans suggests that the peer monitoring thesis is more valid in the traditional bilateral segment than in that via central counterparties, which in fact were created precisely in order to attenuate counterparty risk.

6. Central bank refinancing and customer loans

In analogy with interbank positions, one might argue that in order to show that central bank liquidity injections are effective even with regard to customer credit, the variable Loans should depend on and not cause the banks' demand for central bank refinancing. In this vein, I run a new inverse regression with Loans to the economy as the dependent variable and the central bank refinancing as the key regressor.

This exercise has the merit of 'consolidating' my test of the liquidity hoarding hypothesis, verifying whether the hypothesis holds for the banking system as a whole. To this point, I have shown that banks lend to one another, hence that the liquidity hoarding hypothesis does not hold

among banks. But it could still hold for the relationship between the entire banking system and the rest of the economy. This new exercise shows that this is not the case (Table 11). Central bank refinancing turns out to prompt Loans to the economy: (i) both instrumenting the central bank loans (with its lagged figures) and not; (ii) taking Loans as a ratio to total assets (specification 28); and (iii) using its annual growth rate (specification 29).

7. Robustness checks

I further verified the robustness of the results in several ways.²⁴

7.1 Different estimation methods: difference-in-difference and Tobit models

Table 12 presents the determinants of central bank refinancing using (i) a difference-indifference estimation (instead of the sample time splitting) to control for the impact of the crisis; (ii) a *tobit*-IV (instead of the *ordinary*-IV and the SUR) as regression model; and (iii) lagged interbank positions (see Section 7.2).

Regarding the impact of the crisis, so far I have used a sample time splitting, (repeating the same estimations before and after the onset of the crisis). In the difference-in-difference framework, I consider the crisis as a treatment event, and analyse its effects on all the variables. Equations 1-2 are supplemented by inclusion of an interaction term between the same regressors and a time-dummy variable c_t that takes the value of 1 during the crisis and 0 before.²⁵

Regarding the regression models, the Tobit model is well suited to one of the key-variables (central bank refinancing) because it is continuous and has a constrained range. In fact, central bank refinancing is zero for a substantial part of the sample population as my data refer to all banks in Italy, including those that never directly access it; this provides complementary inferential information and avoids biased sample selection. It is worth emphasizing that the tobit model is run in its IV version.²⁶

Remarkably, these estimation changes do not alter the results at all, either in statistical significance or in the magnitude of marginal effects. In particular, interacted with the dummy variable c_t capturing the crisis phase, Domestic Extra-Group Credits and Net Position are again significantly positive.

²⁴ Since results always remained very similar to those reported in Tables 4-11, for brevity I limit the use of additional tables, but all the robustness checks are available from the author upon request.

²⁵ On a similar use of the dif-in-dif approach, see for example Cetorelli and Strahan (2006).

 $^{^{26}}$ In this framework, I carried out a Wald test of the null hypothesis of no endogeneity. The null was not rejected for all the interbank market segments (interacted with the dummy variable c_t). Things are more differentiated for the variables not interacted with the dummy, which however are not the focus of the diff-in-diff. However, as a check, I also instrumented for the interaction term adding a further equation. By analogy with the other estimations, in Table 12 the instrumented variables are, alternatively, the three Domestic Extra-Group positions.

7.2 Different lags and the persistence of interbank market positions

I ran checks using $x_{i,t-3}$ instead of $x_{i,t}$. This specification makes for better control of endogeneity but estimates a longer-term impact, while liquidity choices tend to be made at very short maturities. In any case, in Table 12 the interbank positions are lagged by one quarter and the results remain equivalent. This long-lasting effect is likely to depend on the persistence of interbank positions. I also verified this persistence empirically in two ways. First, I ran a probit model in which the dependent variable was the share of banks changing total net interbank position compared with the previous period. The estimated share was very low, around 3 per cent during the crisis. Second, I estimated a dynamic panel including the lagged interbank positions as regressors, which always proved to be highly significant.

Likewise, since my estimations compute the regressors in the matrixes $K_{i,t-1}^{R}$ and $K_{i,t-1}^{I}$ as lagged by one period (to avoid new endogeneity in estimating $x_{i,t}$, and to replicate the publication delay needed for mutual assessment by banks), longer lags were used as a robustness check. Again the results remained stable, probably because of the persistence of interbank positions.

7.3 Cooperative banks and branches of foreign banks

A set of checks was run on cooperative banks and branches of foreign banks, which are considered to be unlike other credit institutions. In particular, since I analyse the Eurosystem's liquidity provision, which is decentralized, foreign banks could influence the results if they massively exploit the option to refinance at a given central bank. However, the results remain unchanged when both types of bank are dropped either in turn or jointly. Since the basic results hold even when foreign banks are excluded, this means that the liquidity redistribution towards Cross-Border interbank segments is carried out also by Italian banks. Moreover, since in my framework the number of observations is too small to repeat my exercises only on the two types of banks, I estimated the basic specifications adding the impact of two dummies, for cooperative and foreign banks (but renouncing the fixed effects b_i). This check suggests some observations on the role played by foreign banks, but this calls for specific research. In estimating central bank refinancing, the dummy for foreign banks tends to be positive, both before and after the crisis. The marginal effects indicate that the economic impact is negligible before but sizable after the onset of the crisis, reconfirming that international banking groups raise funds in a decentralised manner.

7.4 Controlling for the endogeneity of other covariates

To verify the stability of the explanatory variables and test for possible collinearity, I adopted two methods: (i) discarding each of the regressors in turn; and (ii) using the IV estimator

for Loans and Fundraising, with a single or a multiple IV estimator. As a vector of instruments, I used the same regressors computed with a two-quarter lag. The results were again confirmed.

7.5 Changing starting dates and periods

In addition to time fixed effects, to test the sensitivity of my results to different dates and periods, I employed two kinds of check. First, I experimented with starting dates other than August 2007 (bringing it forward by one or two months, and postponing it by one to four months); other dates tested were September 2008 (the Lehman Brothers failure), and October 2008 (introduction of the Eurosystem full allotment procedure).²⁷ Second, I tested the stability of the results of the precrisis sample period, which is much longer, juxtaposing two periods of equal length (that is, comparing the last 49 months prior to the critical point with the 49-month-long post-crisis period). In all cases, the results remain substantially stable.

7.6 Changing definitions of variables

Further, I defined some variables in a different way. First of all, I focused on my key variable 'central bank refinancing'. As noted, in my basic estimations, central bank refinancing is measured as *gross* loans. In several checks, I re-measured it as *net* loans, subtracting (from the gross loans that the central bank grants to each bank) the amounts that each bank re-deposits at the central bank. The results do not change. However, I preferred to use the gross variable because deposits at the central bank (i) are typically very low in Italy, even during the crisis; and (ii) as they are basically driven by the euro-area reserve requirement, their inclusion is inconsistent with the variable Fundraising, which is worth keeping because it provides very interesting results.

Then, I focused on three interrelated explanatory variables: Loans, Bad Loans, and Securitized Loans. In the estimations, I separated Loans and Bad Loans from Securitized Loans in order to isolate the effect of the latter (which are more likely to be used as collateral), and at the same time to specifically investigate the pure effect of Loans and Bad Loans (which otherwise could reflect, at least partially, the effect of securitizations). On the other hand, measuring Loans and Bad Loans net of all securitizations decreases their level without reducing credit granted. I verified the results of these variables in three ways. First, I eliminated the variable Securitized Loans and reassigned them as appropriate to either Loans or Bad Loans. Second, I split the variable Securitized Loans between derecognized and non-derecognized loans; attributing the former to

²⁷ Furthermore, since the Bank of Italy's new prudential supervisory reports went into effect as of December 2008, which could have produced some discontinuities in my time-series, I repeated all estimations of my post-crisis period starting from that month onwards.

Loans (and to Bad Loans), and leaving the latter as Securitized Loans.²⁸ Third, I added nonderecognized loans to Loans (and Bad Loans), and left derecognized loans as Securitized Loans. The results never change, probably because the signs of the three variables are identical, both before and after the crisis. My approach definitely demonstrates the positive relationship between central bank liquidity and the variable Loans even net of securitizations.

Finally, to assess the effect of capital adequacy I adopted different proxies as checks. I calculated the numerator of the ratio as either capital and reserves or mandatory capital, and the denominator as either total assets or risk-weighted assets. The results always stand confirmed.

8. Conclusions

Since the outbreak of the financial crisis, liquidity and the functioning of interbank markets have been causes of concern and have been at the centre of the academic and policy debate. This paper contributes to the debate by investigating the determinants and the interrelations between the two main wholesale markets for liquidity: central bank refinancing (the primary liquidity market) and the various segments of the interbank market (the secondary liquidity market).

The paper features several distinctive characteristics. It studies the determinants and the effects of the crisis on central bank refinancing and interbank market jointly. It investigates both the casual directions of their mutual relationship, controlling constantly for mutual endogeneity. It examines the relationships between the two wholesale liquidity markets as well as between those markets, bank loans to the economy and retail bank liquidity, controlling for bank-specific characteristics. The analysis bears on a major central bank, the Eurosystem, and the banking system of a major country, Italy. It distinguishes among the different segments of the interbank market and uses data on all the banks operating in Italy, including those that never directly accessed central bank liquidity, over the period from January 1999 to August 2011. It uses banking group data, reflecting the fact that the decision to access central bank liquidity is likely to be made at group level, and utilizes a broad range of robustness checks.

The analysis does not provide support to the widely held liquidity hoarding hypothesis that during periods of crisis the interbank market ceases to function correctly and central bank injections of liquidity are useless because banks simply build up their liquidity reserves rather than

²⁸ Securitized loans are said to be "derecognized" when they are deleted from the balance sheet of the originator bank because there is a complete transfer of risks, costs, and benefits. Since the breakdown between derecognized and non-derecognized securitized loans is not available from banks for all my sample period, I extended my time series using a bank-level estimation obtained at the Bank of Italy. Likewise, since the Bank of Italy's statistical reports went into effect as of June 2010, and the adoption of the new criteria implied the re-recognition of loans that had previously been removed from the balance sheet, with a corresponding increase in the stock of loans, I restored the continuity of my time series by using the same estimations.

redistributing it to other banks or the real economy. The literature postulates that banks may decide to hoard liquidity and the interbank market may freeze up, but while a part of the literature claims that central bank interventions are ineffective, the prevailing literature recognizes that they are warranted. This paper tests these two conflicting views extensively, and all the results show clearly and robustly that in Italy the interbank markets functioned properly even during the financial crisis, and that the central bank's liquidity circulated among banks and reached the economy.

Further research should quantify the impact of the Italian sovereign debt crisis since the summer of 2011, with its significant fall in the value of government bonds, which are typically used in the interbank market, on the banks' demand for central bank liquidity. Another issue that repays further investigation is the role of foreign banks in cross-border demanding and redistributing central bank liquidity.

References

- Acharya V.V. and Skeie, D., 2011. "A Model of Liquidity Hoarding and Term Premia in Inter-Bank Markets." CEPR Discussion Paper No. 8705.
- Acharya, V., Gale, D., and Yorulmazer T., 2009. "Rollover Risk and Market Freezes", New York University working paper, http://www.nyu.edu/econ/user/galed/papers/paper09-08-31.pdf.
- Acharya, V.V., Gromb, D., and Yorulmazer, T., 2008. "Imperfect Competition in the Interbank Market for Liquidity as a Rationale for Central Banking." Working Paper, London Business School.
- Acharya, V.V., Merrouche, O., 2010. "Precautionary Hoarding of Liquidity and Inter-Bank Markets: Evidence from the Sub-prime Crisis." NBER Working Paper No. 16395.
- Adrian, T. and Shin, H.S., 2009. "Financial intermediation and monetary economics." *Federal Reserve Bank* of New York Staff Reports, 398.
- Affinito, M., 2012. "Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy." *Journal of Banking and Finance*, 36.
- Afonso, G., Kovner, A., Schoar, A., 2011. "Stressed not frozen: The Fed funds market in the financial crisis." The Journal of Finance 66, 1109 1139.
- Albertazzi, U., Ropele, T., Sene G., and F. Signoretti, 2012. "The impact of the sovereign crisis on the lending activity of Italian banks." Bank of Italy, Occasional Paper No.133.
- Allen, F. and E. Carletti, 2008. "The Role of Liquidity in Financial Crises". Jackson Hole Symposium on Maintaining Stability in a Changing Financial System, August 21-23, 2008.
- Allen, F., Carletti, E., and Gale, D., 2009. "Interbank market liquidity and central bank intervention." Journal of Monetary Economics 56, 639–652.
- Angelini, P., Nobili A., and M.C. Picillo, 2011. "The Interbank Market after August 2007: What Has Changed, and Why?", Journal of Money, Credit and Banking, vol. 43(5).
- Armantier, O., Ghysels, E., Sarkar, A., Shrader, J., 2011. "Stigma in Financial Markets: Evidence from Liquidity Auctions and Discount Window Borrowing during the Crisis." Federal Reserve Bank of New York, Staff Report no. 483.
- Ashcraft, A., McAndrews J. and D. Skeie, 2009. "Precautionary Reserves and the Interbank Market." Federal Reserve Bank of New York Staff Reports, 370.
- Ashcraft, A.B., Bech, M.L., Frame W.S., 2008. "The Federal Home Loan Bank System: The Lender of Next-to-Last Resort?" Federal Reserve Bank of New York, Staff Report no. 357.
- Bank of Italy, 2011a. Economic Bulletin No. 61, July.
- Bank of Italy, 2011b. Financial Stability Report No. 2, November.
- Bank of Italy, 2012. Economic Bulletin No. 69, July.
- Berger A.N. and C.H.S. Bouwman, 2009. "Bank Liquidity Creation", The Review of Financial Studies, 9.
- Berrospide J. 2012. "Liquidity Hoarding and the Financial Crisis: An Empirical Evaluation." Federal Reserve System, mimeo.
- Bhattacharya, Sudipto, and Douglas Gale, 1987. "Preference Shocks, Liquidity and Central Bank Policy." In *New Approaches to Monetary Economics*, edited by William A. Barnett and Kenneth J. Singleton, pp. 69–88. New York: Cambridge University Press.
- Bindseil, U., Nyborg, K.G., and Strebulaev I.A. 2009. "Repo Auctions and the Market for Liquidity." Journal of Money, Credit and banking, Vol. 41-7, 1391-1421.
- Bofondi, M., L. Carpinelli and E. Sette, 2013. "Credit supply during a sovereign debt crisis." Bank of Italy Working Paper 909.
- Borio, C. and Zhu, H., 2008. "Capital regulation, risk-taking and monetary policy: A missing link in the transmission mechanism?" *Bank for International Settlements Working Paper*, 268.
- Breitung, J. and Nautz, D., 2001. "The empirical performance of the ECB's repo auctions: evidence from aggregated and individual bidding data." Journal of International Money and Finance 20, 839–856.
- Brunetti C., di Filippo M., Harris JH., 2011. "Effects of Central Bank Intervention on the Interbank Market During the Sub-Prime Crisis," Review of Financial Studies, 24-6.
- Brunnermeier, M., 2009. "Deciphering the 2007-08 Liquidity and Credit Crunch." Journal of Economic Perspectives 23, 77-100.
- Bruno, G., Ordine, M., and Scalia A., 2005. "Banks' Participation in the Eurosystem Auctions and Money Market Integration." Banca d'Italia Discussion Paper No. 562.

- Cappelletti G., A. De Socio, G. Guazzarotti, E. Mallucci, 2011. "The impact of the financial crisis on interbank funding: evidence from Italian balance sheet data." Questioni di Economia e Finanza, 95, Banca d'Italia.
- Cassola, N., Hortacsu, A., Kastl, J., 2011. "The 2007 Subprime Market Crisis Through the Lens of European Central Bank Auctions for Short-Term Funds." ECB, Working Paper, 1374.
- Cecioni M., G. Ferrero, A. Secchi, 2011. "Unconventional monetary policy in theory and in practice." Questioni di Economia e Finanza, 102, Banca d'Italia.
- Cetorelli N. and P.H. Strahan, 2006. "Finance as a Barrier to Entry: Bank Competition and Industry Structure in Local U.S. Markets", The Journal of Finance, 61-1.
- Christensen J.H.E., Lopez J.A., Rudebusch G.D., 2009. "Do Central Bank Liquidity Facilities Affect Interbank Lending Rates?" Federal Reserve Bank of San Francisco, Working paper series 13.
- Craig, B., and F. Fecht, 2007. "The Eurosystem Money Market Auctions: A Banking Perspective." *Journal* of Banking and Finance, 31, 2925–44.
- Diamond, D. and Rajan, R., 2005. "Liquidity Shortages and Banking Crises." Journal of Finance 60, 615–647.
- Diamond, D. and Rajan, R., 2008. "Illiquidity and Interest Rate Policy." Working paper, University of Chicago.
- Dinger V. and J. von Hagen, 2009. "Does Interbank Borrowing Reduce Bank Risk?" Journal of Money, Credit, and Banking, 41,2-3, 491-506.
- ECB, 2011. "Target2 balances of National Central Banks in the Euro-area." ECB Monthly Bulletin, October, 35-40.
- ECB, 2012. "Euro Money Market Study."
- Edlin A.S., Jaffee D.M., 2009. "Show Me The Money." The Economists' Voice, Berkeley Electronic Press, 6(4).
- Ennis, H.M., and J.A. Weinberg, 2009. "A model of stigma in the fed funds market." Working Paper Federal Reserve Bank of Richmond.
- Eser F., M. Carmona Amaro, M. Iacobelli, M. Rubens, 2012. "The use of the Eurosystem's monetary policy instruments and operational framework since 2009." ECB Occasional Papers, 135.
- Fecht, F., Nyborg, K.G., Rocholl, J., 2011. "The price of liquidity. The effects of market conditions and bank characteristics." ECB Working Paper Series 1376.
- Financial Times, 2008. "Hoarding by banks stokes fear over crisis", 26 March and 9 November.
- Flannery, M.J., 1996. "Financial crises, payment system problems, and discount window lending." Journal of Money, Credit, and Banking 28, 804-824.
- Freixas, X. and C. Holthausen, 2005. "Interbank Market Integration under Asymmetric Information." *Review of Financial Studies*.
- Freixas, X., Giannini, C., Hoggarth, G., Soussa, F., 1999. "Lender of last resort: a review of literature." Financial Stability Review, 151-167.
- Freixas, X., Martin, A., Skeie, D., 2011. "Bank Liquidity, Interbank Markets, and Monetary Policy." Review of Financial Studies, 24-8.
- Freixas, X., Parigi, B., Rochet, J.C., 2000. "Systemic risk, interbank relations, and liquidity provision by the central bank." Journal of Money, Credit and Banking 32, 611-638.
- Furfine C.H., 2001. "Banks as Monitors of Other Banks: Evidence from the Overnight Federal Funds Market." *The Journal of Business*, 74 (1), 33–58.
- Furfine C.H., 2003. "Standing facilities and Interbank borrowing: Evidence from Federal Reserve's new Discount Window." *International Finance* 6, 329-347.
- Furfine C.H., 2009. "The Liquidity of Liquid Markets during the Financial Crisis." Mimeo.
- Gilbert, R.A., 1995. "Determinants of federal reserve lending to failed banks." Journal of Economics and Business 47 (5), 397–408.
- Goodfriend, M.R. and King, R.G., 1988. "Financial deregulation, monetary policy, and central banking." Federal Reserve Bank of Richmond Economic Review 74 (3), 3–22.
- Gorton, G. B., and A. Metrick, 2011. "Securitized Banking and the Run on Repo." Journal of Financial Economics, 104, 425–451.
- Heider, F., Hoerova M., Holthausen, C., 2009. "Liquidity hoarding and interbank market spreads: The role of counterparty risk." In: CEPR-CREI-Journal of 1080 Financial Intermediation Conference on the Financial Crisis, Universitat 1081 Pompeu Fabra.

- Kashyap, A.K., Rajan, R., Stein, J.C., 2002. "Banks as liquidity providers: an explanation for the coexistence of lending and deposit-taking." Journal of Finance 57, 33 {73.
- Kaufman, G.G., 1991. "Lender of last resort: A contemporary perspective." Journal of Financial Services Research 5, 95–110.
- Keister, T. and McAndrews, J., 2009. "Why are banks holding so many excess reserves?" Federal Reserve Bank of New York, Staff Report 380.
- King T.B., 2008. "Discipline and Liquidity in the Interbank Market." Journal of Money, Banking and Credit., 40, 2-3, 295-317.
- Linzert, T., Dieter Nautz, and Jorg Breitung, 2006. "Bidder Behavior in Central Bank Repo Auctions: Evidence from the Bundesbank." *Journal of International Financial Markets, Institutions, and Money*, 16, 215–50.
- Linzert, Tobias, Dieter Nautz, and Ulrich Bindseil, 2007. "Bidding Behavior in the Longer Term Refinancing Operations of the European Central Bank: Evidence from a Panel Sample Selection Model." *Journal of Banking and Finance*, 31, 1521–43.
- Martin, A., D. R. Skeie, and E.-L. von Thadden, 2010. "Repo Runs." FRB of New York Staff Report No. 444.
- McAndrews, J., A. Sarkar and Z. Wang, 2008. "The Effect of the Term Auction Facility on the London Inter-Bank Offered Rate." Staff Reports 335, Federal Reserve Bank of New York.
- Michaud F.L. and Upper C., 2008. "What Drives Interbank Rates? Evidence from the Libor Panel." BIS Quarterly Review.
- Nautz, D., and J. Oechssler, 2003. "The Repo Auctions of the European Central Bank and the Vanishing Quota Puzzle." *Scandinavian Journal of Economics*, 105, 207–20.
- Nyborg, K.G. and Strebulaev, I.A., 2004. "Multiple unit auctions and short squeezes." Review of Financial Studies 17, 545-580.
- Nyborg, K.G., U. Bindseil, and I.A. Strebulaev, 2002. "Bidding and Performance in Repo Auctions: Evidence from ECB Open Market Operations." ECB Working Paper No. 157.
- Peristiani S., 1998. The growing reluctance to borrow at the discount window: An empirical investigation. *The Review of Economics and Statistics*, 611-620.
- Schnabl, P., 2012. "The International Transmission of Bank Liquidity Shocks: Evidence from an Emerging Market." Journal of Finance, 67, 897-932.
- Schwarz C., 2009. "Mind the Gap: Distentangling Credit and Liquidity in Risk Spreads." Working paper, University of Pennsylvania Wharton School of Business.
- Stojanovic D., Vaughan M.D., and Yeager T.J., 2008. "Do Federal Home Loan Bank membership and advances increase bank risk-taking?" *Journal of Banking and Finance*, 32, 680-698.
- The Economist, 2007. "Central banks struggle to prevent money markets from drying up." Aug 16th.















Figure 3. Italian interbank market segments: Debts as percentage shares of bank total assets



	Key variables (scaled by total assets)		Obs	Mean	Sd. Dev.	Min	Max
	Loans form central bank		61,196	0.009	0.005	0.000	0.132
		Debts	61,196	0.029	0.021	0.000	0.110
	Domestic Extra-Group	Credits	61,196	0.036	0.041	0.000	0.220
		Net	61,196	0.003	0.048	-0.110	0.200
	Domestic Infra-Group	Debts or Credits	61,196	0.037	0.022	0.000	0.389
		Debts	61,196	0.058	0.034	0.000	0.756
Interbank	Cross-Border Extra-Group	Credits	61,196	0.049	0.010	0.000	0.149
market		Net	61,196	-0.015	0.013	-0.190	0.199
sections		Debts	61,196	0.001	0.005	0.000	0.146
	Cross-Border Infra-Group	Credits	61,196	0.002	0.003	0.000	0.065
		Net	61,196	-0.001	0.004	-0.134	0.031
		Debts	61,196	0.0040	0.001	0.000	0.074
	Central Counterparties	Credits	61,196	0.0038	0.001	0.000	0.051
		Net	61,196	-0.0002	0.001	-0.068	0.051

Table 1. Summary statistics of key variables

Table 2. Relations among key variables

		Loans from central	Domes	stic Extra	-Group	Domestic Infra-	Cross-B	order Extr	a-Group	Cross-B	order Infr	a-Group	Centra	al Counter	parties
		bank	Debts	Credits	Net	Group	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net
Loans from central	bank	1													
	Debts	0.0931*	1												
Domestic Extra-Group	Credits	-0.0612*	-0.1799*	1											
	Net	-0.0912*	-0.5752*	0.9016*	1										
Domestic Infra-Group	Debts or Credits	0.1392*	0.1054*	-0.0809*	-0.1125*	1									
	Debts	0.0848*	0.1499*	-0.0516*	-0.1088*	0.1715*	1								
Cross-Border Extra-Group	Credits	0.1460*	0.1875*	0.0170*	-0.0676*	0.3809*	0.5112*	1							
	Net	-0.0961*	-0.1914*	0.0695*	0.1399*	-0.1710*	-0.6615*	0.0311*	1						
	Debts	0.1546*	0.0976*	-0.0411*	-0.0759*	0.4840*	0.2214*	0.4568*	-0.2263*	1					
Cross-Border Infra-Group	Credits	0.1534*	0.1119*	-0.0238*	-0.0672*	0.2594*	0.2448*	0.4450*	-0.2878*	0.6476*	1				
I	Net	-0.0970*	-0.0511*	0.0372*	0.0530*	-0.4520*	-0.1220*	-0.2942*	0.0995*	-0.8590*	0.1661*	1			
	Debts	0.1086*	0.0130*	-0.0436*	-0.0422*	0.2443*	0.0469*	0.0715*	-0.0027	0.1273*	0.2367*	0.0230*	1		
Central Counterparties	Credits	0.1141*	0.0292*	-0.0358*	-0.0425*	0.2400*	0.0745*	0.1286*	0.0063	0.1235*	0.1874*	-0.0079	0.5071*	1	
	Net	-0.0432*	0.006	0.0245*	0.0181*	-0.1092*	-0.0726*	-0.0960*	0.0149*	-0.0818*	-0.1804*	-0.0350*	-0.7872*	0.1323*	1

***, **, and * denote statistical significance at 1, 5 and 10 % level.

Table	3.	Summary	statistics	of ex	planatory	variables
		\sim		· · · ·	prover j	

	Name	Definition	Obs	Mean	Sd. Dev.	Min	Max
	Size	Log (Total assets)	61,196	5.674	1.650	1.386	13.662
	Loans	Total performing (non-securitized) loans to the domestic private sector / Total assets	61,196	0.559	0.137	0.003	0.790
	Bad Loans	Total non-performing (non-securitized) loans (private sector) / Total performing (non-securitized) loans (private sector)	61,196	0.046	0.049	0.000	0.300
Matrix K ^R	Portfolio of Government Debt Securities	Holdings of Euro-area Government bonds / Total assets	61,196	0.022	0.006	0.000	0.150
banks' characteristics/	Portfolio of Bank Bonds	Holdings of their own bonds and of other banks' bonds / Total assets	61,196	0.025	0.029	0.000	0.160
regressors	Securitized Loans	Total (dereconized and non-dereconized) securitized loans / Total assets	61,196	0.010	0.027	0.000	0.220
	ROE	Net profits / (Capital and reserves)	61,196	0.007	0.029	-0.048	0.140
	Capital	Regulatory capital / Total risk weighted assets	61,196	0.124	0.037	0.068	0.339
	Fundraising	(Total deposits and bonds) / Total assets	61,196	0.732	0.087	0.000	0.961
Matrix $K^{I}_{i,t-1}$:	Rating	Rating agency scores	61,196	7.724	1.309	2	11
instruments	Banks without rating (0-1)	Banks without rating (0-1)	61,196	0.587	0.199	0	1

Table 4.Determinants of central bank refinancing

Results of equation 1. Sample time splitting: each specification is identically repeated before and after the crisis. Dependent variable $y_{i,i}$: ratio of total gross loans from central bank to total assets. Estimation method: *ordinary*-IV and SUR. Endogenous and instrumented set of regressors $x_{i,i}$: in Specifications (1)-(3): Domestic-Extra-Group positions; in Specifications (4)-(6): Central Couterparties positions. Specifications (7)-(9) report the SUR results of the first equation of a system of three equations (the other two equations refer to both Domestic Extra-Group and CCPs positions). Endogenous regressors are crossed out when not instrumented and unreliable. Corresponding IV first stage results, and results of the other two equations are reported in Table 7.

Dependent	t variable:								Lo	ans from	central bo	ınk							
Samp	le period:				Pre-	crisis po	eriod				i I I			Post	-crisis p	eriod			
Estimatio	n method:		IV (1)			IV (2)			SUR		l I	IV (1)			IV (2)			SUR	
Spec	ifications:	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Debts	-0.0426***			0.000			0.055			-0.022			-0.066***			0.480		
Domestic Extra Group	Credits	0.002	-0.0412***		0.001	0.000		0.000	-0.021**			0.618***		0.007	0.005		0.700	0.208*	
Extra-Oroup		-	0.012	-0.0668***		0.000	0.002***		0.009	-0.034**	 	0.222	0.269*		0.004	0.0204***		0.125	0.865**
	Net			0.019			0.001			0.016			0.138			0.002			0.376
Domostia Infra Casur	Debts or	-0.0170***	-0.0215***	-0.0301***	-0.015***	-0.018***	-0.010***	-0.007	-0.021***	-0.019***	-0.015	-0.121**	-0.019	-0.020	-0.112	-0.003	-0.096**	-0.084**	-0.246*
Domestic Infra-Group	Credits	0.003	0.004	0.004	0.002	0.002	0.002	0.007	0.004	0.004	0.014	0.059	0.018	0.022	0.030	0.032	0.037	0.041	0.133
	Dahta	-0.001			0.003			0.003			-0.03***			-0.0298***			-0.0512***		
	Debis	0.002			0.003			0.006			0.004			0.004			0.010		
Cross-Border	Cuadita		-0.00673**			-0.010			-0.005**		 	-0.132***			-0.581***			-0.497***	
Extra-Group	Creans		0.003			0.025			0.003			0.041			0.117			0.095	
	Nat			-0.0226***			-0.016***			-0.022***	İ		0.129***			0.0848***			0.227***
	Net			0.006			0.002			0.005			0.027			0.010			0.076
	Dahta	-0.018***			-0.006			0.003			-0.005			-0.015			-0.172		
	Debis	0.005			0.004			0.016			0.028			0.044			0.180		
Cross-Border	Custite		0.0195**			0.0250**			0.026***			0.798**			0.364*			0.341*	
Infra-Group	Creatis		0.010			0.008			0.009		Ì	0.315			0.177			0.207	
	Nat			0.0199**			0.0243***			0.023***	I I		0.187**			0.038			0.016
	Net			0.008			0.007			0.007	1		0.083			0.060			0.138
	DL	-0.142***			0.382			0.528			0.225***			0.175			-0.845		
	Debts	0.052			0.241			0.814			0.036			0.222			0.583		
Counterparties	Credits		-0.146***			0.464**			0.063			0.222**			-0.815***			-0.564***	
Counterparties			0.030	9.022		0.168	0.126**		0.048	0.168*		0.108	010/***		0.189	0 355		0.141	0 272**
	Net			0.079			0.048			0.092			0.065			0.376			0.162
Dependent variable		Do	mestic Extra-G	roup	Cen	tral Counterpa	rties	Do. and C	mestic Extra-Gr 'entral Counterp	oup parties	Do	mestic Extra-Gr	oup	Cen	ntral Counterpa	rties	Doi and C	nestic Extra-Gi entral Counter	oup parties
in the first stage:		Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

(to be continued)

Dependent variable:								Lo	ans from	central ba	nk							
Sample period:				Pre-	crisis pe	eriod				 			Post	-crisis p	eriod			
Estimation method:		IV (1)			IV (2)			SUR		Î	IV (1)			IV (2)			SUR	
Specifications:	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Size	-0.001***	0.000	0.000	-0.001***	-0.001***	-0.001***	-0.0005***	-0.0004***	0.000	0.002	0.000	0.008**	0.003***	0.002	0.002**	-0.003	0.003**	0.019**
5120	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.003	0.002	0.003	0.001	0.001	0.001	0.008	0.001	0.008
Loong	-0.005***	-0.009***	-0.010***	0.001	0.002	0.002	0.001	-0.003*	-0.006*	0.000	0.20***	0.095**	0.001	0.003	0.011***	-0.009	0.069*	0.298**
Loans	0.001	0.001	0.003	0.000	0.000	0.000	0.003	0.002	0.003	0.003	0.071	0.042	0.002	0.003	0.002	0.013	0.041	0.128
D II	0.000	-0.0049***	0.000	0.002	0.002	0.002	0.004	0.000	-0.001	0.012	0.15***	0.045***	0.009*	0.053***	0.024***	0.034***	0.077***	0.123**
Bad Loans	0.001	0.002	0.002	0.000	0.001	0.001	0.004	0.001	0.001	0.008	0.046	0.015	0.003	0.011	0.004	0.011	0.026	0.049
Portfolio of Government	0.0137***	-0.011	-0.008	0.015***	0.019***	0.015***	0.014***	0.007	-0.006	-0.0615***	0.157	0.026	-0.071***	-0.073**	-0.034*	-0.021	0.004	0.190
Debt Securities	0.002	0.006	0.007	0.002	0.003	0.005	0.002	0.005	0.006	0.019	0.098	0.042	0.014	0.027	0.015	0.030	0.057	0.130
Portfolio of	-0.00986***	-0.002	-0.004	0.000	0.003	-0.011	0.008	0.002	-0.005**	-0.003	0.151***	0.007	-0.009	0.037	0.004	0.085**	0.081***	-0.018
Bank Bonds	0.004	0.001	0.003	0.001	0.001	0.001	0.014	0.001	0.003	0.018	0.058	0.006	0.004	0.011	0.004	0.032	0.029	0.015
	-0.0023***	-0.004***	0.000	-0.002**	-0.004***	0.000	-0.0009	-0.003***	0.000	0.106***	0.175***	0.121***	0.103***	0.168***	0.122***	0.177***	0.173***	0.090***
Securitized Loans	-0.001	-0.001	-0.001	0.001	0.001	0.001	0.001	0.001	0.001	-0.013	-0.029	0.009	0.007	0.017	0.007	0.026	0.018	0.020
	0.000	0.000	0.000	0.001	0.000	0.000	0.000	0.000	0.000	-0.001	0.008	0.003	-0.001	0.001	0.000	0.000	0.002	0.003
ROE	0.001	0.001	0.001	0.001	0.001	0.001	0.002	0.001	0.001	0.002	0.006	0.003	0.002	0.003	0.002	0.004	0.004	0.005
	-0.0172***	-0.0146***	-0.00996***	-0.008***	-0.009***	-0.016***	-0.001	-0.011***	-0.012***	-0.0306**	0.056	-0.0456**	-0.036***	-0.057***	-0.028***	0.030	-0.024	-0.124**
Capital	0.003	0.002	0.003	0.001	0.001	0.001	0.012	0.001	0.002	0.015	0.034	0.015	0.006	0.015	0.008	0.025	0.034	0.046
	-0.0122***	-0.000973*	0.006	-0.002***	-0.002***	-0.009***	0.006	-0.002***	0.002	-0.028	-0.0705***	-0.135***	-0.039***	0.004	-0.024***	-0.102**	-0.021	-0.396**
Fundraising	0.004	0.001	0.005	0.000	0.000	0.001	0.015	0.000	0.004	0.029	0.022	0.040	0.003	0.006	0.003	0.048	0.013	0.164
	0.021***	0.017***	0.005	0.014***	0.015***	0.016***	0.003	0.013***	0.009**	-0.019	-0 179***	-0.095***	-0.025*	-0.026	-0.033***	-0.0016	-0.096**	-0.211***
Constant	0.003	0.001	0.004	0.001	0.001	0.002	0.014	0.001	0.004	0.016	0.033	0.031	0.009	0.017	0.011	0.018	0.036	0.082
Bank fixed effects	ves	ves	ves	ves	ves	ves	ves	ves	ves	ves	Ves	ves	ves	ves	ves	ves	ves	ves
Time fixed effets	900	900	900	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	900	900	900	900	905	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	905	905	905	905	905	900		<i>J</i> e 0
	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Number of observations	44,641	44,526	44,029	43,544	43,323	43,027	44,336	44,145	43,852	16,545	16,459	16,343	16,545	16,459	16,343	16,466	16,417	16,292
Dependent variable	Dor	mestic Extra-Gi	oup	Cen	tral Counterpa	rties	Don and C	nestic Extra-Gr entral Counterp	oup parties	Don	nestic Extra-Gr	oup	Cen	tral Counterpa	rties	Dor and C	nestic Extra-Gr entral Counter	oup parties
in the first stage:	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net

Table 4. Determinants of central bank refinancing (continued)

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level

Depe	endent variable:		Loan	s from	central	bank	
	Sample period:	Pre	-crisis pe	riod	Post	-crisis pe	riod
Esti	mation method:	IV (1)	IV (2)	SUR	IV (1)	IV (2)	SUR
	Specifications:	1-3	4-6	7-9	1-3	4-6	7-9
	Debts	-0.2	ns	ns	ns	-0.2	ns
Domestic Extra-Group	Credits	-0.5	15	-0.3	3.7	<u> </u>	2.9
	Net	-0.4	0.0	-0.4	2.1	1.9	2.0
Domestic Infra-Group	0	-0.1	-0.1	-0.1	-0.1	ns	-0.1
	Debts	ns	ns	ns	-0.4	-0.4	-0.5
Cross-Border Extra-Group	Credits	-0.1	ns	-0.1	-0.1	-0.6	-0.5
	Net	0.0	-0.1	-0.1	0.1	0.1	0.1
	Debts	-0.1	ns	ns	ns	ns	ns
Cross-Border Infra-Group	Credits	0.1	0.1	0.1	0.4	0.2	0.3
	Net	0.1	0.1	0.1	0.2	ns	ns
	Debts	-0.5	ns	ns	2.2	ns	ns
Central Counterparties	Credits	-05	0.5	ns	0,1	-1.0	-1.1
	Net	ns	1.0	0.2	-1.3	ns	-1.4
Size		-0.2	-0.2	-0.2	1.6	1.2	1.5
Loans		-0.3	ns	-0.2	3.5	3.2	4.4
Bad Loans		-0.1	ns	ns	0.4	0.2	0.5
Portfolio of Government Debt	Securities	0.1	0.1	0.1	-0.1	-0.1	ns
Portfolio of Bank Bond	ds	-0.1	ns	-0.1	0.5	ns	0.3
Securitized Loans		-0.1	-0.1	-0.1	0.1	0.2	0.2
ROE		ns	ns	ns	ns	ns	ns
Capital		-0.1	-0.2	-0.2	-0.3	-0.3	-0.5
Fundraising		-0.2	-0.2	-0.2	-1.8	-1.4	-2.3

Table 5.Determinants of central bank refinancing

Marginal effects, averaged across the specifications, of the estimations of Table 4.

Table reports marginal effects, averaged across the specifications, of all estimations shown in Table 4. The marginal effects quantify the estimated economic impact of each regressor on the dependent variable 'central bank refinancing', other things being equal. The estimated effect of each determinant is computed as the change in the percentage share of the total loans from central bank to total assets between the 25th to the 75th percentile of each variable. Like in Table 4, endogenous but non-instrumented regressors are crossed out because unreliable. ns denotes statistically non-significant regressors.

Table 6.Determinants of central bank refinancing

Results of equation 1. Sample time splitting: only post-crisis results are reported. Dependent variable $y_{i,t}$ ratio of total gross loans from central bank to total assets. Estimation method: *ordinary*-IV. Endogenous and instrumented set of regressors $x_{i,t}$: in Specifications 10-12: sum of Domestic-Extra-Group + CCPs positions; in Specifications 13-15: sum of Domestic Extra-Group + CCPs + Cross-Border Extra-Group + Cross-Border Infra-Group positions. Corresponding IV first stage results are not reported because substantially equivalent to those of Table 7 as for Domestic Extra-Group segment.

Dependent	variable:		Lo	ans from	central ban	ık	
Sample	e period:			Post-cris	sis period		
Estimation	method:		IV (3)			IV (4)	
Specif	ications:	(10)	(11)	(12)	(13)	(14)	(15)
	Debts	0.040 0.085					
Domestic Extra-Group + Central Counterparties	Credits		0.692** <i>0.258</i>				
	Net			0.154** 0.071			
Domestic Extra-Group + Central	Debts				0.022 0.044		
Counterparties + Cross-Border Extra-Group + Cross-Border Infra-	Credits					0.546 *** 0.178	
Group	Net						0.681 * 0.353
Domestic Infra-Group	Debts or Credits	-0.022 0.019	0.105 0.068	-0.042 0.063	-0.016 0.018	0.082 0.055	0.101 0.070
	Debts	-0.032*** 0.004					
Cross-Border Extra-Group	Credits		-0.106*** 0.033				
	Net			0.097*** 0.019			
	Debts	-0.029 0.033					
Cross-Border Infra-Group	Credits		0.992*** 0.350				
	Net			0.175*** 0.062			
Size		0.001	0.000	0.006***	0.001	-0.001	-0.014
Loong		-0.001	0.202**	0.002	-0.002	0.182 ***	0.268 **
Loans		0.002	0.082	0.026	0.002	0.058	0.100
Bad Loans		0.015**	0.138*** 0.043	0.033*** 0.008	0.016 ***	0.172 *** 0.041	-0.027 0.048
Portfolio of Covernment Debt See	ritios	-0.052***	0.153	-0.009	-0.082 ***	0.106	-0.484 **
Fortiono of Government Debt Secu	intes	0.015	0.141	0.028	0.020	0.086	0.214
Portfolio of Bank Bonds		0.007	0.152**	0.007 0.004	0.007 0.008	0.256 ***	0.022
		0.151***	0.191***	0.120***	0.112 ***	0.191 ***	0.082 **
Securitized Loans		0.007	0.029	0.007	0.005	0.024	0.039
ROE		0.000	0.008	0.002	0.000	0.006	-0.010
		-0.002	0.006	-0.032***	-0.02	0.005	0.008
Capital		0.011	0.047	0.009	0.008	0.024	0.033
Fundraising		-0.019	-0.082***	-0.067**	-0.015	-0.056 ***	-0.309 *
T unurarsing		0.017	0.025	0.032	0.012	0.016	0.175
Constant		-0.289** 0.113	-0.377*** 0.020	-0.185*** 0.025	-0.051 ** 0.020	-0.274 *** 0.059	0.095 0.089
Bank fixed effects		yes	yes	yes	yes	yes	yes
Time fixed effets		yes	yes	yes	yes	yes	yes
Number of observations		16,466	16,417	16,292	16,466	16,417	16,292
Dependent variable in the first sta	age:	Domestic	Extra-Group Counterpartie	+ Central s	Domestic E Cross-Borde	xtra-Group + r Extra and Ir	CCPs + nfra Group
Corrossending	first star	Debts	Credits	Net	Debts	Credits	Net

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

Table 7.Determinants of interbank market positions

Results of equation 2. Table 7 couples with Table 4 (i.e. it contains the corresponding IV first stage results, and results of the second and third equation of the SUR estimation of a system of three equations). Sample time splitting: the specifications of Table 7, repeated before and after the crisis, correspond to the specifications of Table 4. Set of dependent variables $x_{i,t}$: Specifications (1)-(3): Domestic-Extra-Group positions; Specifications (4)-(6): Central Couterparties positions; Specifications (7)-(9): both Domestic-Extra-Group and CCPs positions. Estimation method: *ordinary*-IV and SUR. Corresponding IV second stage results, and results of the first equation of the SUR estimation are reported in Table 4.

Dep	pendent variable:	Domes	tic Extra-	Group	Centra	l Counter	parties	Domestic Extra-Group	Central Counterp.	Domestic Extra-Group	Central Counterp.	Domestic Extra-Group	Central Counterp.	Domes	stic Extra-	Group	Centra	l Counter	parties	Domestic Extra-Group	Central Counterp.	Domestic Extra-Group	Central Counterp.	Domestic Extra-Grou	Central p Counterp.
		Debts	Credits	Net	Debts	Credits	Net	De	bts	Crea	lits	N	let	Debts	Credits	Net	Debts	Credits	Net	D	ebts	Cre	edits		Net
	Sample period:]	Pre-cris	is perio	d									I	Post-cris	sis perio	bd				
Es	timation method:		IV (1)			IV (2)				SU	R			1	IV (1)			IV (2)				S	UR		
	Specifications:	(1)	(2)	(3)	(4)	(5)	(6)	C	7)	(8)	(9	9)	(1)	(2)	(3)	(4)	(5)	(6)	((7)	(8)		(9)
Loons from cont	wal bank							-0.648***	-0.059	-0.323***	0.693	0.132***	0.137***	1						0.144	-0.104*	0.198***	-0.178***	0.336***	-0.269***
Loans from cent	IT AT DAILK							0.239	0.078	0.057	0.582	0.040	0.049	 						0.165	0.060	0.068	0.054	0.121	0.060
	Dehts				0.006***				0.115***					1			0.008***				0.502**				
	Debis				0.001				0.016					!			0.001				0.190				
Domestic	Credits					0.000					0.204			ł				0.000					0.037		
Extra-Group						0.000					0.304			!				0.000					0.029		
	Net						0.000						0.058						0.0021***						0.232***
							0.000						0.035	i !					0.001						0.046
Domestic Infra-	Debts or Credits	-0.127***	-0.318***	-0.159***	0.006***	0.0069***	0.003***	-0.039	0.009***	0.131	0.167	0.007	0.028**	-0.066***	-0.418***	-0.058	-0.0971***	-0.0216***	0.0716***	-0.007	-0.099***	-0.108	-0.015***	0.152	0.066***
Group	1	0.007	0.015	0.016	0.000	0.000	0.000	0.032	0.002	0.088	0.179	0.058	0.012	0.029	0.046	0.049	0.003	0.002	0.003	0.041	0.030	0.319	0.005	0.393	0.009
	Debts	-0.092***			0.0010***			-0.073***	0.0072***					0.015*			-0.0103***			0.0627**	-0.053**				
		0.008			0.000			-0.008	0.001					0.009			0.001			0.023	0.023				
Cross-Border	Credits		-0.168***			0.0067***				-0.120***	0.045			1	0.097			-0.0716***				0.312	-0.088***		
Extra-Group			0.022			0.000				0.044	0.053			ł	0.086			0.002				0.773	0.008		
	Net			-0.318***			0.000					-0.115**	0.033**	!		-0.50'/***			-0.0068***					-0.448	-0.061***
		0.146***		0.013	0.0020***		0.000	0.006***	0.020***			0.054	0.015	0.029		0.037	0.175***		0.002	0.024	0.104***			0.514	0.014
	Debts	0.021			0.001			-0.035	0.020***					-0.038			-0.175***			0.034	-0.194				
Cross Bandan		0.021	-0.165*		0.001	0.002		-0.055	0.005	-0 118***	-0.173			0.002	-1 430**		0.010	-0.012		0.070	0.059	-0.206	0.061		
Infra-Group	Credits		0.090			0.002				0.024	0.135			1	0.513			0.030				0.128	0.041		
				-0.111*			-0.001					-0.479***	-0.031**			-0.456***			-0.1385***					-0.828	0.004
	Net			0.064			0.001					0.139	0.013	ļ		0.158			0.009					0.944	0.037
	Debts	0.529***												0.518***											
Central	<i>G</i> . It.	0.152	-0.411											0.095	0.070										
Counterparties	Credits		0.256	0.446										 	0.264	0.42046.5									
	Net			0.446 0.436										İ		0.420*** 0.121									

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

(to be continued)

		unes ()1 III.	ci bai	111 1110	II ILUU	Post	ions (001111	in cicci	/													
Dependent variable:	Domes	tic Extra	-Group	Centra	l Counter	parties	Domestic Extra- Group	Central Counterp.	Domestic Extra- Group	Central Counterp.	Domestic Extra- Group	Central Counterp.	Domes	tic Extra	-Group	Centra	l Counter	rparties	Domestic Extra- Group	Central Counterp.	Domestic Extra- Group	Central Counterp.	Domestic Extra- Group	Central Counterp.
	Debts	Credits	Net	Debts	Credits	Net	De	ebts	Cre	edits	Λ	let	Debts	Credits	Net	Debts	Credits	Net	De	ebts	Cre	edits	Λ	√et
Sample period:		•	•		Р	re-cris	is perio	d	•		•				•		Р	ost-cris	sis peri	od				
Estimation method:		IV (1)			IV (2)				S	UR			Ì	IV (1)			IV (2)				SI	UR		
Specifications:	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
~ ~ ~	0.011**	0.013***	0.024***	-0.0005***	-0.0003***	-0.000***	0.0036**	-0.0001	0.0365***	-0.0024**	0.023***	0.000	0.020***	0.004	-0.022***	0.0009***	0.000	-0.0010***	0.017***	0.007*	0.004	0.0000***	-0.035	-0.0052***
Size	0.005	0.001	0.002	0.000	0.000	0.000	0.001	0.000	0.005	0.001	0.003	0.000	0.001	0.003	0.004	0.000	0.000	0.000	0.002	0.003	0.021	0.000	0.238	0.046
	0.029***	-0.175***	-0.218***	-0.0003***	-0.0004***	0.000	0.023***	0.002	-0.205***	0.030	-0.218***	0.010	0.027***	-0.332***	-0.348***	-0.0021***	0.000	0.0030***	0.026***	-0.0157***	-0.332***	0.012	-0.340***	0.080***
Loans	0.001	0.003	0.003	0.000	0.000	0.000	-0.001	-0.001	0.007	0.050	0.006	0007	0.003	0.007	0.008	0.001	0.000	0.001	0.003	0.006	0.018	0.009	0.024	0.015
Pad Loons	-0.039***	-0.116***	-0.076***	-0.0002**	0.000	0.000	-0.051***	-0.0047***	-0.180***	0.010	-0.081***	0.001	-0.061***	-0.198***	-0.098***	-0.001	0.0043***	0.0053***	-0.069***	0.0358**	-0.229**	0.014**	-0.150	0.033***
Dau Loans	0.003	0.006	0.005	0.000	0.000	0.000	-0.005	-0.001	0.015	0.023	0.010	0.001	0.007	0.014	0.017	0.001	0.001	0.001	0.009	0.006	0.088	0.006	0.094	0.006
Portfolio of	-0.009	-0.472***	-0.329***	0.000	-0.001	-0.001**	-0.099***	0.000	-0.990***	-0.0171	-0.385***	0.011	-0.112***	-0.380***	-0.255***	-0.004	0.000	0.006	-0.044	0.002	-0.262	0.007	-0.103	0.050***
Government Debt Securities	0.012	0.029	0.029	0.000	0.001	0.000	0.036	-0.001	0.106	0.053	0.044	0.010	0.024	0.054	0.067	0.005	0.003	0.004	0.049	0.037	0.524	0.011	0.302	0.017
Portfolio of	-0.160***	-0.091***	0.154***	0.000	0.000	0.000	-0.162***	-0.018***	-0.186***	-0.0051	0.293***	0.006**	-0.148***	-0.240***	-0.004	0.010	0.004	-0.0026**	-0.154***	0.039***	-0.246***	0.014**	-0.011	-0.0048
Bank Bonds	0.004	0.010	0.010	0.000	0.000	0.000	-0.004	-0.002	0.025	0.008	0.046	-0.003	0.006	0.015	0.017	0.001	0.001	0.001	0.007	0.029	0.033	0.006	0.043	0.003
Securitized Loans	-0.014***	-0.037***	-0.011	0.0010***	0.0015***	0.001***	-0.005	0.007	-0.038**	0.021	-0.018	0.000	-0.102***	-0.109***	-0.023	0.0190***	0.0071***	-0.0106***	-0.260***	0.134**	-0.328	0.031***	-0.404	-0.024***
	0.004	0.008	0.010	0.000	0.000	0.000	-0.005	0.000	0.019	0.021	0.012	-0.001	0.010	0.022	0.025	0.001	0.001	0.001	0.073	0.080	0.753	0.008	0.823	0.008
ROE	0.001	0.007	0.004	0.000	0.000	0.000		0.000		-0.0024		0.000	-0.005	-0.011	-0.010	0.0012*	0.000	-0.0011*		0.006		0.000		0.000
ROL	0.004	0.010	0.011	0.000	0.000	0.000		0.000		0.008		0.000	0.004	0.007	0.008	0.001	0.000	0.001		0.000		0.000		0.000
Canital	-0.127***	-0.088***	0.120***	0.000	-0.0004**	0.000*	-0.082***	-0.017***	-0.241***	0.080	0.344***	0.016***	-0.116***	-0.122***	0.088***	0.0043**	-0.0061***	-0.0115***	-0.074***	0.031	-0.089	-0.001	0.161	-0.033***
Capitai	0.005	0.011	0.010	0.000	0.000	0.000	0.022	-0.003	0.059	0.078	0.069	-0.005	0.012	0.027	0.031	0.002	0.001	0.002	0.025	0.022	0.184	-0.004	0.203	0.007
Fundraising	-0.187***	0.011**	0.258***	0.000	0.000	0.000	-0.148***	-0.019***	0.0912***	0.013	0.365***	0.002	-0.236***	0.099***	0.410***	0.0089***	0.0025***	-0.0052***	-0.222***	0.107	0.123	0.000	0.471***	-0.106***
i unui aising	0.002	0.004	0.006	0.000	0.000	0.000	0.007	-0.003	0.016	0.010	0.036	-0.003	0.003	0.007	0.010	0.001	0.000	0.001	0.016	0.139	0.108	-0.003	0.120	0.020
Rating	0.003***	0.002**	-0.001	-0.0003***	-0.0005***	-0.000***	0.003***		0.0075***		0.000		0.006***	0.003	-0.011**	0.0041***	0.0006**	-0.0025***	0.008***		0.013		-0.014***	
	0.000	0.001	0.001	0.000	0.000	0.000	0.000		0.002		0.001		0.002	0.006	0.005	0.000	0.000	0.000	0.002		0.013		-0.004	
Banks without Rating	-0.022***	0.002	0.020***	0.0019***	0.0030***	0.001***	-0.018***		-0.001		0.013*		-0.066***	-0.057	0.079**	-0.0331***	-0.003	0.0219***	-0.070***		-0.090		0.108**	
	0.002	0.006	0.006	0.000	0.000	0.000	-0.003		-0.006		0.008		0.015	0.046	0.039	0.002	0.002	0.002	0.016		0.060		0.053	
Constant	0.417***	-0.030	-0.612***	0.004	0.007	0.002***	0.038	0.019***	-0.509***	-0.095	-0.452***	-0.010**	-0.410***	0.214***	0.680***	-0.033	-0.002	0.026	0.053**	-0.001	-0.025	-0.017*	0.447***	0.0565***
	0.006	0.139	0.015	0.000	0.000	0.000	0.034	0.003	0.094	0.086	0.078	0.004	0.020	0.041	0.053	0.003	0.002	0.003	0.021	0.022	0.266	0.009	0.264	0.015
Bank fixed effects	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Time fixed effets	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Number of observations	44,641	44,526	44,029	43,544	43,323	43,027	44	,336	44	,145	43,	,852	16,545	16,459	16,343	16,545	16,459	16,343	16	,466	16	,417	16	,292

Table 7. Determinants of interbank market positions (continued)

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

Table 8.Determinants of interbank market positions

Results of equation 1. Sample time splitting: each specification is identically repeated before and after the crisis. Estimation method: *ordinary*-IV. Dependent variable $y_{i,t}$: Specifications (16)-(18): Domestic-Extra-Group positions; Specifications (19)-(21): sum of Domestic-Extra-Group + CCPs + Cross-Border Extra-Group + Cross-Border Infra-Group positions. Endogenous and instrumented regressor $x_{i,t}$: ratio of total gross loans from central bank to total assets. Corresponding IV first stage results are not reported because equivalent to those of Table 4.

Dependen	t variable:	Dome: (scale)	stic Extra- d by total d	Group 1ssets)	Dom Counterpar + 0 (sci	estic Extra + rties + Cross Cross-Borde aled by total	Central s-Border Extra r Infra assets)	Dome (scale	stic Extra- d by total o	Group assets)	Domes Counterpo Extra + (scale	tic Extra + 0 arties + Cro. Cross-Bord ed by total as	Central ss-Border er Infra ssets)
		Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net
San	ple period:		Р	re-cri	sis per	iod			Po	ost-cris	sis peri	od	
Estimat	ion method:		IV (5)			IV (6)		 	IV (5)	-		IV (6)	
Spe	ecifications:	(16)	(17)	(18)	(19)	(20)	(21)	(16)	(17)	(18)	(19)	(20)	(21)
Loans from central ba	nk	-0.049 ***	-0.108 **	0.344 ***	-0.518 ***	-0.222 ***	0.242 ***	-0.350 ***	0.191 *	0.463 ***	-0.436 ***	0.079 **	0.512 ***
	Debts or	-0.120 ***	-0.320 ***	-0.159 ***	-0.243 ***	-0.405 ***	-0.137 ***	-0.062 ***	-0.194 ***	-0.032	-0.376 ***	-0.270 ***	0.041
Domestic Infra-Group	Credits	0.007	0.016	0.018	0.008	0.016	0.018	0.021	0.041	0.048	0.027	0.042	0.048
	Debts	-0.077 ***						0.003					
		0.008	-0.159 ***					0.009	0.089				
Cross-Border Extra-Group	Credits		0.025						0.055				
	Net			-0.279 ***				i I		-0.237 ***			
		-0.142 ***		0.017				-0.038		0.033			
	Debts	0.024						0.060					
Cross-Border Infra-Group	Credits		-0.146 *					1	-0.129 **				
eross border mind eroup			0.090	0.121 **				 	0.052	0 452 ***			
	Net			0.064				i		-0.432 *** 0.141			
	Debts	0.494 ***						0.248 ***					
	Debis	0.160						0.045					
Central Counterparties	Credits		-0.400					i I	0.095				
			0.270	0.457				 	0.154	0.303 ***			
	Net			0.477				 		0.117			
Size		0.000	0.012 ***	0.014 ***	-0.001	0.013 ***	0.016 ***	0.024 ***	0.005 *	-0.024 ***	0.041 ***	0.007 **	-0.029 ***
		0.000	0.001	0.001	0.001	0.001	0.001	0.001	0.003	-0.331 ***	0.002	0.003	0.003
Loans		0.001	0.003	0.003	0.001	0.003	0.003	0.003	0.007	0.008	0.004	0.007	0.008
Bad Loans		-0.039 ***	-0.118 ***	-0.069 ***	-0.052 ***	-0.123 ***	-0.061 ***	-0.059 ***	-0.205 ***	-0.112 ***	-0.095 ***	-0.202 ***	-0.097 ***
Deutfelie of		0.002	0.006	0.006	0.003	0.006	0.007	0.007	0.014	0.016	0.009	0.014	0.017
Government Debt Secu	rities	-0.003	-0.470 ***	-0.326 ***	0.050 ***	-0.447/***	-0.326 ***	-0.147/***	-0.420 ***	-0.230 ***	-0.274 ***	-0.418 ***	-0.574 ***
Deutfelie of Deule Deu		-0.166 ***	-0.085 ***	0.154 ***	-0.214 ***	-0.087 ***	0.200 ***	-0.152 ***	-0.237 ***	0.006	-0.181 ***	-0.258 ***	0.025
Portiolio of Bank Bor	ids	0.004	0.010	0.010	0.005	0.010	0.011	0.007	0.015	0.017	0.010	0.015	0.017
Securitized Loans		-0.013 ***	-0.028 ***	-0.014	-0.052 ***	-0.022 **	0.012	-0.068 ***	-0.111 ***	-0.076 ***	0.004	-0.109 **	-0.128 ***
		0.004 -0.010 **	0.008	0.009	0.004 -0.010 *	-0.008	-0.009	0.010	-0.021	-0.009	-0.003	-0.022	-0.025
ROE		0.004	0.010	0.011	0.005	0.010	0.011	0.003	0.007	0.008	0.005	0.007	0.008
Canital		-0.140 ***	-0.079 ***	0.130 ***	-0.200 ***	-0.088 ***	0.161 ***	-0.122 ***	-0.127 ***	0.069 **	-0.169 ***	-0.117 ***	0.061 **
Cupitur		0.005	0.011	0.012	0.005	0.011	0.012	0.012	0.026	0.030	0.017	0.027	0.031
Fundraising		-0.169	0.013	0.257 ***	-0.214 ***	0.012 **	0.297 ***	-0.256 ***	0.091 ***	0.423 ***	-0.298	0.085	0.451 ***
Dating		0.003 ***	0.002 ***	-0.002	0.005	0.002 **	-0.004 ***	0.007 ***	0.004	-0.010 **	0.000	-0.002	-0.004
Katilig		0.000	0.001	0.001	0.000	0.001	0.001	0.002	0.005	0.005	0.003	0.004	0.004
Banks without Ratin	g	-0.021 ***	0.001 ***	0.021 ***	-0.036 *	0.006	0.038 ***	-0.069 ***	-0.057	0.086 **	-0.058 ***	-0.015	0.059
		0.002	0.000	-0.189 ***	0.005	0.000	-0.267 ***	-0.145 ***	0.045	0.309 ***	-0.169 ***	0.032	0.037
Constant		0.006	0.015	0.017	0.007	0.015	0.017	0.021	0.046	0.049	0.028	0.043	0.050
Bank fixed effects		yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Time fixed effets		yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
Dependent variable in the first	t stage	44,330	44,140	43,832	44,330	44,145 L.08	ns from o	entral ^{10,400}	ank	10,292	10,400	10,417	10,292
Corresponding first stage results (Loans fro	m central bank	l c instrumente	ed hv lagged	figures) are	not reported	1 1.00			Juin				

Corresponding first stage results (Loans from central bank instrumented by lagged figures) are not reported.

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

Dependent variable:		Domestic Extra-Group (scaled by total assets)			Central Counterparties (scaled by total assets)		Domestic Extra + Central Counterparties + Cross- Border Extra + Cross- Border Infra (scaled by total assets)		Domestic Extra-Group (scaled by total assets)		Central Counterparties (scaled by total assets)			Domestic Extra + Central Counterparties + Cross- Border Extra + Cross- Border Infra (scaled by total assets)					
		Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net	Debts	Credits	Net
Sample period:					Pre	-crisis pe	eriod							Pos	t-crisis pe	riod			1
Estimatio	on method:	IV(1); SUR;	IV(5)	г	V(2); SU	R	IV(6)			IV(1); SUR; IV(5)			IV(2); SUR			IV(6)		
Spec	cifications:	1; 7; 16	2; 8; 17	3; 9; 18	4; 7	5; 8	6; 9	19	20	21	1; 7; 16	2; 8; 17	3; 9; 18	4; 7	5; 8	6; 9	19	20	21
Loans from central bank		-0.5	-1.1	3.4	ns	ns	0.0	-2.2	-1.8	2.4	-3.5	1.8	4.6	-0.1	-0.2	-0.4	-4.4	0.6	5.1
	Debts	na			0.0						na			0.1					
Domestic Extra-Group	Credits		na			ns						na			ns				
	Net			na			ns						na			0.1			
Domestic Infra-Group	Debts or Credits	0.0	-0.1	0.0	0.0	0.0	0.0	0.0	-0.1	-0.1	0.0	0.0	ns	-0.1	0.0	0.1	-0.1	-0.1	0.0
	Debts	-0.8			0.1						0.2			-1.5					
Cross-Border Extra-Group	Credits		-0.1			0.1						ns			-0.5				
	Net			0.0			0.1						0.0			0.0			
	Debts	-0.1			0.0						ns			-1.5					
Cross-Border Infra-Group	Credits		-0.1			ns						-0.7			ns				
	Net			-0.1			0.0						-0.4			-1.3			
	Debts	1.0			na						1.6			na					
Central Counterparties	Credits		ns			na						ns			na				
	Net			ns			na						2.7			na			
Size		1.1	2.5	2.8	-0.2	-0.4	-0.1	ns	2.7	3.1	4.2	2.2	-4.8	2.2	0.3	-2.3	5.4	1.5	-5.9
Loans		4.9	-4.2	-5.0	-0.1	-0.1	ns	0.8	-4.3	-5.0	0.6	-7.8	-7.8	-0.5	ns	0.7	0.8	-8.0	-7.8
Bad Loans		-0.3	-0.8	-0.3	0.0	ns	ns	-0.3	-0.6	-0.3	-0.3	-1.0	-0.5	0.1	0.2	0.2	-0.5	-1.0	-0.5
Portfolio of Government Debt Sect	urities	-0.1	-0.3	-0.2	ns	ns	0.0	0.0	-0.3	-0.2	-0.1	-0.2	-0.9	ns	ns	0.3	-0.2	-0.3	-0.3
Portfolio of Bank Bonds		-0.5	-0.3	0.4	-0.1	ns	0.1	-0.6	-0.3	0.6	-0.5	-0.7	ns	0.2	0.3	-0.2	-0.5	-0.8	ns
Securitized Loans		0.0	0.0	ns	0.0	0.0	0.0	-0.1	0.0	ns	-0.1	-0.1	-0.1	0.2	0.1	-0.1	ns	-0.1	-0.1
ROE		0.0	ns	ns	ns	ns	ns	0.0	ns	ns	0.0	0.0	ns	0.0	ns	0.0	ns	ns	ns
Capital		-0.6	-0.4	0.6	ns	0.0	0.0	-1.0	-0.4	0.8	-0.6	-0.7	0.4	0.2	-0.2	-0.5	-0.8	-0.6	0.3
Fundraising		-3.9	0.2	3.7	-0.2	ns	ns	-3.2	0.2	4.5	-3.7	1.5	6.2	1.4	0.4	-0.9	-4.5	1.3	6.8
Rating		0.6	0.4	ns	-0.2	-0.3	-0.2	ns	0.3	-0.7	1.4	ns	-2.3	1.3	0.8	-0.7	ns	ns	ns
Banks withou Rating		-2.4	ns	2.0	0.8	1.0	0.7	-3.6	ns	3.8	-6.8	ns	8.6	-3.1	ns	1.3	-5.8	ns	ns

Table 9.Determinants of interbank market positionsMarginal effects, averaged across estimation models and specifications, of the estimations of Tables 7 and 8.

Table reports marginal effects, averaged across estimation models and specifications, of the estimations shown in Tables 7 and 8. The marginal effects quantify the estimated economic impact of each regressor on the dependent variable, other things being equal. The estimated effect of each determinant is computed as the change in the percentage share of interbank positions to total assets between the 25th to the 75th percentile of each variable. ns denotes statistically non-significant regressors; na non-applicable regressors.

Table 10.Determinants of interbank market positions

Results of equation 1. Sample time splitting: each specification is identically repeated before and after the crisis. Estimation method: *ordinary*-IV. Dependent variable $y_{i,i}$: Specifications (22)-(24): Domestic-Extra-Group positions; Specifications (25)-(27): sum of Domestic-Extra-Group + CCPs + Cross-Border Extra-Group + Cross-Border Infra-Group positions. Compared to Table 8, dependent variables are computed as month growth rates. Endogenous and instrumented regressor $x_{i,i}$: ratio of total gross loans from central bank to total assets. Corresponding IV first stage results are not reported because equivalent to those of Table 4.

Dapandant variabla:		Domo	atia Entra	Crown	Dom	estic Extra +	Central Bandan Fatan	Domo	otio Entra	Crown	Domestic Extra + Central Counterparties + Cross-Border Extra + Cross-Border Infra			
		Dome. (g	rowth rate	es)	+ (Cross-Border	-Boraer Extra Infra	Dome. (g	rowth rate	es)				
Dependen	i variabie.	Debts	Credits	Net	Debts	(growin rate Credits	Net	Debts	Credits	Net	(. Debts	growin rates Credits	Net	
Sar		P	re_cri	sis ner	ind			р	ost-cri	sis neri	od			
Estimation method:			IV (7)	10-011	IV (8)				IV (7)	051-011	IV (8)			
Spe	ecifications:	(22)	(23)	(24)	(25)	(26)	(27)	(22)	(23)	(24)	(25)	(26)	(27)	
Loans from central ba	nk	-0.537 ***	-0.511	-6.288	-0.312	-0.420	-2.562	-2.519	4.610 ***	4.361 *	-5.677 **	6.040 ***	2.911	
	uik	0.097	3.357	7.092	1.063	0.904	2.550	2.439	1.030	2.553	2.447	1.174	1.963	
Domestic Infra-Group	Debts or Credits	-1.225 **	-4.425 ***	-1.466	-1.217 *	-1.385 **	9.698 ***	-2.405	-2.229 *	-1.353	-3.978	-3.399 **	4.572 **	
		-3.699 ***	0.050	1.900	0.705	0.571	1.000	-0.737	1.177	4.277	2.755	1.524	2.017	
	Debts	0.721						1.173						
Cross-Border Extra-Group	Credits		0.422						1.444					
Cross Dorder Extra Group	creaus		1.065						1.596					
	Net			-2.068						-16.35 ***				
		-1 306		1.93/				2 717		2.898				
	Debts	1.982						7.825						
Course Develop Infra Course	C I'		-2.493						2.483					
Cross-Border Infra-Group	Creatts		3.612						14.962					
	Net			-18.74 **						-27.02 **				
		7 706		7.096				2.262		12.447				
Central Counterparties	Debts	13.46						5.205						
		15.70	3.571					2.000	0.088 *					
	Credits		11.097						0.046					
	Net			-3.829 ***						-3.055				
	1101			0.526						10.323				
Size		0.021	0.108 **	-0.031	0.077	0.218 ***	0.002	0.140	0.180 **	-0.688 **	0.344	0.316 ***	0.435	
		-0.197	-0.614 ***	0.141	-0.142	-0.888 ***	-0.493	-0.467	-1 572 ***	0.509	0.262	-1 266 ***	1.427	
Loans		0.136	0.119	0.367	0.158	0.107	0.301	0.592	0.202	0.714	0.579	0.211	3.227	
Pad Loons		-0.666 *	-0.230	0.308	0.569	-0.376 *	0.212	-0.311	-1.043 **	-1.002	-0.309	-1.456 ***	-0.435	
Bau Loans		0.376	0.238	0.735	0.437	0.217	0.620	1.244	0.417	1.484	1.265	0.444	6.828	
Portfolio of		0.517	-0.864	-4.484	-4.044 **	-4.100 ***	-2.737	-7.445	-5.829 ***	9.840 *	-10.952 **	-2.984 *	-0.773	
Government Debt Secu	rities	1.270	0.212	3.561	1.477	1.062	3.016	4.707	1.685	6.003	4.778	1.826	27.901	
Portfolio of Bank Bon	nds	0.442	0.390	0.245	0.501	-1.099	0.997	1.183	0.429	-1.985	-1.082	-1.090	-0.824	
0		-1.353 ***	-0.625 *	2.024 *	0.274	0.427	-1.414	-1.313	0.451	7.380 ***	2.057 ***	1.458 **	3.777 ***	
Securitized Loans		0.348	0.343	1.060	0.411	0.313	0.879	1.423	0.618	2.152	1.462	0.682	1.040	
ROE		-1.117 **	0.409	2.849 **	0.131	-0.315	0.313	0.188	0.063	0.083	-0.432	-0.284	0.892	
		0.434	0.403	1.233	0.506	0.366	1.036	0.596	0.208	0.729	0.618	0.229	3.517	
Capital		0.309	0.269	-1.670	-1.784 ***	-1.590 ***	-1.624 ***	4.343 *	0.886	0.472	4.909 **	-1.162	10.840	
		-1 011 ***	0.430	-0.285	-1 997 ***	-0.593	-1 203	-2.576 ***	0.227	1 851 **	-3 009 ***	-0.513	-0.666	
Fundraising		0.243	0.209	0.644	0.264	0.825	0.916	0.667	0.236	0.840	0.619	0.350	3.825	
Dating		0.036	-0.103 ***	0.359 ***	-0.038	-0.005	-0.113	0.026	-0.071	-0.360	0.043	-0.061	-0.849	
Katilig		0.032	0.035	0.107	0.038	0.032	0.091	0.243	0.157	0.400	0.256	0.124	1.898	
Banks without Rating		-0.237	0.413 **	-2.913 ***	0.390	0.084	0.577	-0.127	0.276	2.576	-0.452	0.321	6.790	
	-	0.207	0.229	0.693	0.243	0.209	0.582	2.053	1.290	3.311	2.120	1.031	15.713	
Constant		0.505	-0.203	-2.008 1.885	0.334	-1.410 ···· 0.546	1.545 ***	-0.110	-1.055	-1.808 -	-2.392 3.542	-2.377 "	-7.599 20.455	
Bank fixed effects		yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	
Time fixed effets		yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	
Number of observations		42,115	41,767	40,472	42,115	41,767	40,472	15,653	15,588	15,451	15,653	15,588	15,451	
Dependent variable in the first	st stage:					Loa	ans from	central	bank					
				<i>c</i> >										

Corresponding first stage results (Loans from central bank instrumented by lagged figures) are not reported.

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

Table 11. Reverse causality between central bank refinancing and Loans (to bank customers) Sample time splitting: only post-crisis results are reported. Estimation method: *ordinary*-IV. Dependent variable: Specification (28): Loans as ratios to total assets; Specification (29): Loans as year growth rate. Endogenous and instrumented regressor: ratio of total gross loans from central bank to total assets. Corresponding IV first stage results are not reported because equivalent to those of Table 4.

	Loans						
Dependent variable		scaled by total assets	growth rate				
Sam	Post-crisis period						
Estimati	on method:	IV(9)	IV(10)				
Spe	(28)	(29)					
Loans form central ba	0.204 ***	0.598 *					
	0.038	0.351					
Domestic Extra-Group	Net	-0.294 ***	-0.118				
		0.007	0.141				
Domestic Infra-Group	Debts or	-0.429 ***	-4.652 ***				
	Credits	0.046	0.920				
Cross-Border Extra-Group	Net	-0.034	4.139 ***				
		0.031	0.591				
Cross-Border Infra-Group	-0.433 ***	7.722 ***					
_	0.133	2.604					
Central Counterparties	0.613	50.269 ***					
	0.110	2.114					
Size	-0.063 ***	0.514					
	0.003	0.064					
Bad Loans	-0.799	-0.930					
Portfolio of	-0.475 ***	3.086 **					
Government Debt Secur	-0.475	1 257					
	0.036 **	0.211					
Portfolio of Bank Bon	0.016	0.307					
	-0.242 ***	-1.386 ***					
Securitized Loans	0.023	0.415					
DOE	-0.006	-0.154					
ROE	0.008	0.149					
Carrital	0.419 ***	2.906 ***					
Capitai	0.029	0.577					
Eundroising		0.319 ***	-0.499 ***				
T undraising		0.009	0.181				
Rating		-0.021 ***	0.238 *				
Katilig		0.004	0.126				
Banks without Ratin	σ	0.147 ***	-2.079 **				
	6	0.035	1.022				
Constant		0.135 ***	-6.295 ***				
		0.005	1.018				
Bank fixed effects	yes	yes					
Time fixed effets		yes	yes				
Number of observations		16,103	15,794				
Dependent variable in the firs	t stage:	Loans from	central bank				
Company on din a finat state as more lta and mot non	ortad						

Corresponding first stage results are not reported

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level.

Table 12. Robustness check: Determinants of central bank refinancing

Results of equation 1. Difference-in-difference estimations. Dependent variable $y_{i,t}$: ratio of total gross loans from central bank to total assets. Estimation method: *tobit*-IV. Endogenous and instrumented set of lagged regressors $x_{i,t-3}$: Domestic-Extra-Group positions. Corresponding IV first stage results are not reported because equivalent to those of Tables 7 and 8. Each specification contains two columns: (*a*) shows the results of the variables interacted with the crisis dummy c_t (representing the real focus of the diff-in-diff); (*b*) refers to the non-interacted regressors.

Dependent	Loans from central bank											
Sampl	All: diff-indiff model											
Estimation		Tobit-IV										
			(30))		(32	Marginal effects			
specifications:		(a)	(́ (b)	(a)		́(b)	(a)	<u> </u>	, (b)	(a)	(b)
	Debts	0.219		-0.234							ns	ns
Domostio		0.314		0.303								
Extra-Group	Credits				0.826 **	le ale	-0.289				1.8	ns
Exau-Group					0.291		0.244	0.728 -	k	-0.462		
	Net							0.443		0.434	2.3	ns
Domestic Infra-Group	Debts or	-0.089 *	**	-0.056 ***	-0.096 **	**	-0.068 ***	-0.058 *	**	-0.069 ***	0.0	0.0
Domestie Inna-Group	Credits	0.022		0.009	0.021		0.009	0.024		0.009	0.0	0.0
	Debts	-0.176 *	**	-0.011							-2.2	ns
Cross-Border		0.025		0.013	0.106 **		0.002				1	
Extra-Group	Credits				-0.196 ***		0.002				-0.2	ns
					0.004		0.022	0.247	k 34 34	-0.001		
	Net							0.034		0.016	0.0	ns
	Debts	-0.059		-0.104 ***							ns	-0.1
		0.064		0.028								
Cross-Border	Credits				0.159 *		-0.223 ***				0.1	-0.1
Infra-Group					0.096		0.053	0.107	k	0.172 ***	1	
	Net							0.056		0.034	0.1	0.2
Central Counterparties	Data	-0.162 *	**	0.133 ***				0.020		0.057	0.2	0.2
	Debts	0.031		0.016							-0.3	0.2
	Credits				-0.092 **	* *	0.035 ***				-0.5	0.2
					0.024		0.010					
	Net							-0.068 *	***	-0.003	-0.4	ns
		-0.001		0.011 ***	0.001 *		0.012 ***	0.018	***	0.009		
Size		0.001		0.000	0.001		0.000	0.001		0.000	0.4	2.3
Loans		0.095 *	**	-0.024 ***	0.116 **	* *	-0.022 ***	0.125	***	-0.022 ***	28	-0.6
Loans		0.011		0.004	0.010		0.005	0.012		0.005	2.0	-0.0
Bad Loans		0.058 *	*	-0.014	0.094 **	**	-0.018 *	0.088 *	***	-0.030 ***	0.5	-0.2
		0.023		0.009	0.022		0.010	0.025		0.011		
Portfolio of Government Debt Se	curities	-0.409 *	т т	0.001	-0.284 **	• •	0.034	-0.106		-0.006	-0.2	ns
		0.139 *	**	0.101 ***	0.112 **	* *	0.123 ***	0.193	***	0.064 ***		
Portiolio of Bank Bonds		0.021		0.015	0.021		0.014	0.022		0.013	0.5	0.4
Securitized Loans		0.248 *	**	-0.097 ***	0.252 **	* *	-0.108 ***	0.242	***	-0.094 ***	03	-0.2
		0.017		0.013	0.016		0.014	0.018		0.015		
ROE		-0.043 *		0.016	-0.062 **	k	0.005	-0.060 *	**	0.013	-0.2	ns
		-0.154 *	**	-0.059 ***	-0.168 **	**	-0.079 ***	-0.109 3	***	-0.130 ***		
Capital		0.029		0.014	0.029		0.015	0.031		0.014	-0.8	-0.5
Eve desigin s		-0.058 *	**	-0.017 **	-0.078 **	**	-0.013 **	-0.029 *	***	-0.075 ***	0.0	0.7
Fundraising		0.008		0.008	0.011		0.005	0.010		0.005	-0.9	-0.7
Constant		-0.1		***	-0.	-0.103 ***		-C	.057			
Constant	0.010			0.007			0.	007	1			
Bank fixed effects	yes			yes				yes	1			
Time fixed effets	yes			yes				yes	1			
Number of observations	59,499 59,191 58,778						/8	L				
Dependent variable in the first	Debts Credits Net											
Corresponding first st	s (eqaution	(eqaution 2) are not reported because analogous to those of Table 7										

Table reports regression coefficients and associated standard errors in italics. ***, **, and * denote statistical significance at 1, 5 and 10 % level. The dummy c_t (taking the value of 1 during the crisis and 0 before) is not separately estimated thanks to the presence of the month fixed effects p_t , which in addition allow a better identification. By analogy with Table 4, one could read column (*a*) as the post-crisis outcomes, and column (*b*) as the pre-crisis outcomes. However, the interpretation of interaction-term components' coefficients cannot be the same as if they were ordinary coefficients in a strictly additive model. Table also reports marginal effects, averaged across specifications. The marginal effects quantify the estimated economic impact of each regressor on the dependent variable, other things being equal. The estimated effect of each determinant is computed as the change in the percentage share of interbank positions to total assets between the 25th to the 75th percentile of each variable. ns denotes statistically non-significant regressors.

- N. 905 Family firms and the Great Recession: out of sight, out of mind?, by Leandro D'Aurizio and Livio Romano (April 2013).
- N. 906 Price discovery in the Italian sovereign bonds market: the role of order flow, by Alessandro Girardi and Claudio Impenna (April 2013).
- N. 907 Public-private wage differentials in euro area countries: evidence from quantile decomposition analysis, by Domenico Depalo and Raffaela Giordano (April 2013).
- N. 908 Asking income and consumption questions in the same survey: what are the risks?, by Giulia Cifaldi and Andrea Neri (April 2013).
- N. 909 Credit supply during a sovereign debt crisis, by Marcello Bofondi, Luisa Carpinelli and Enrico Sette (April 2013).
- N. 910 Geography, productivity and trade: does selection explain why some locations are more productive than others?, by Antonio Accetturo, Valter Di Giacinto, Giacinto Micucci and Marcello Pagnini (April 2013).
- N. 911 *Trust and preferences: evidence from survey data*, by Giuseppe Albanese, Guido de Blasio and Paolo Sestito (April 2013).
- N. 912 *Tempered stable Ornstein-Uhlenbeck processes: a practical view*, by Michele Leonardo Bianchi, Svetlozar T. Rachev and Frank J. Fabozzi (June 2013).
- N. 913 Forward-looking robust portfolio selection, by Sara Cecchetti and Laura Sigalotti (June 2013).
- N. 914 When the baby cries at night. Inelastic buyers in non-competitive markets, by Giacomo Calzolari, Andrea Ichino, Francesco Manaresi and Viki Nellas (June 2013).
- N. 915 Local development that money can't buy: Italy's Contratti di Programma, by Monica Andini and Guido de Blasio (June 2013).
- N. 916 The effect of organized crime on public funds, by Guglielmo Barone and Gaia Narciso (June 2013).
- N. 917 *Relationship and transaction lending in a crisis*, by Patrick Bolton, Xavier Freixas, Leonardo Gambacorta and Paolo Emilio Mistrulli (July 2013).
- N. 918 Macroeconomic effects of precautionary demand for oil, by Alessio Anzuini, Patrizio Pagano and Massimiliano Pisani (July 2013).
- N. 919 Skill upgrading and exports, by Antonio Accetturo, Matteo Bugamelli and Andrea Lamorgese (July 2013).
- N. 920 *Tracking world trade and GDP in real time*, by Roberto Golinelli and Giuseppe Parigi (July 2013).
- N. 921 Should monetary policy lean against the wind? An analysis based on a DSGE model with banking, by Leonardo Gambacorta and Federico M. Signoretti (July 2013).
- N. 922 *Marshallian labor market pooling: evidence from Italy*, by Monica Andini, Guido de Blasio, Gilles Duranton and William C. Strange (July 2013).
- N. 923 Do euro area countries respond asymmetrically to the common monetary policy?, by Matteo Barigozzi, Antonio M. Conti and Matteo Luciani (July 2013).
- N. 924 *Trade elasticity and vertical specialisation*, by Ines Buono and Filippo Vergara Caffarelli (July 2013).
- N. 925 Down and out in Italian towns: measuring the impact of economic downturns on crime, by Guido de Blasio and Carlo Menon (July 2013).
- N. 926 The procyclicality of foreign bank lending: evidence from the global financial crisis, by Ugo Albertazzi and Margherita Bottero (July 2013).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- A. PRATI and M. SBRACIA, *Uncertainty and currency crises: evidence from survey data*, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, v. 42, 19-21, pp. 2399-2415, **TD No. 458 (December 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, v. 50, 4, pp. 833-857, **TD No. 591 (May 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area,* International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, *New Eurocoin: Tracking Economic Growth in Real Time*, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, **TD No. 631 (June 2007).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- L. GAMBACORTA and C. ROSSI, *Modelling bank lending in the euro area: a nonlinear approach*, Applied Financial Economics, v. 20, 14, pp. 1099-1112 ,**TD No. 650** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, GIANMARCO I.P. OTTAVIANO and G. PERI, *The labor market impact of immigration on the western german labor market in the 1990s*, European Economic Review, v. 54, 4, pp. 550-570, **TD No. 687 (August 2008).**
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums*, Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, *Dropping the Books and Working Off the Books*, Labour, v. 24, 2, pp. 139-162, **TD No. 702 (January 2009).**
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, v. 14, 5, pp. 677-708, **TD No. 706** (March 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740** (January 2010).
- M. AFFINITO and E. TAGLIAFERRI, *Why do (or did?) banks securitize their loans? Evidence from Italy*, Journal of Financial Stability, v. 6, 4, pp. 189-202, **TD No. 741 (January 2010).**
- S. FEDERICO, Outsourcing versus integration at home or abroad and firm heterogeneity, Empirica, v. 37, 1, pp. 47-63, **TD No. 742 (February 2010).**
- V. DI GIACINTO, On vector autoregressive modeling in space and time, Journal of Geographical Systems, v. 12, 2, pp. 125-154, TD No. 746 (February 2010).

- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, v. 34, 9, pp. 1791-1812, **TD No. 747** (March 2010).
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- P. CIPOLLONE, P. MONTANARO and P. SESTITO, Value-added measures in Italian high schools: problems and findings, Giornale degli economisti e annali di economia, v. 69, 2, pp. 81-114, TD No. 754 (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, *Asset-based measurement of poverty*, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757 (April 2010).**
- S. DI ADDARIO and D. VURI, Entrepreneurship and market size. the case of young college graduates in *Italy*, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**
- A. CALZA and A. ZAGHINI, *Sectoral money demand and the great disinflation in the US*, Journal of Money, Credit, and Banking, v. 42, 8, pp. 1663-1678, **TD No. 785 (January 2011).**

2011

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No. 605 (December 2006).**
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, Directed matching with endogenous markov probability: clients or competitors?, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- R. GIORDANO and P. TOMMASINO, *What determines debt intolerance? The role of political and monetary institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, v. 18, 6, pp. 724-49, **TD No. 732 (November 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union:* the Case of Italy, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., **TD No. 749 (March 2010).**
- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets*, International Journal of Industrial Organization, v. 29, 6, pp. 694-704, **TD No. 760** (June 2010).

- G. BARBIERI, C. ROSSETTI e P. SESTITO, The determinants of teacher mobility: evidence using Italian teachers' transfer applications, Economics of Education Review, v. 30, 6, pp. 1430-1444, TD No. 761 (marzo 2010).
- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, **TD No.** 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- S. FEDERICO and A. FELETTIGH, *Measuring the price elasticity of import demand in the destination markets of italian exports*, Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**
- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807** (April 2011).
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837** (November 2011).
- F. CAPRIOLI, P. RIZZA and P. TOMMASINO, *Optimal fiscal policy when agents fear government default*, Revue Economique, v. 62, 6, pp. 1031-1043, **TD No. 859** (March 2012).

2012

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate*, International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis*, Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704 (March 2009).**
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010).**
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, v. 86, 2, pp. 269-283, **TD No. 743 (February 2010).**
- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**

- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, Journal of Chinese Economic and Business Studies, v. 10, 3, pp. 275-299, **TD No. 838 (November 2011).**
- P. GUERRIERI and F. VERGARA CAFFARELLI, Trade Openness and International Fragmentation of Production in the European Union: The New Divide?, Review of International Economics, v. 20, 3, pp. 535-551, TD No. 855 (February 2012).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

2013

- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, **TD No. 815 (July 2011).**
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**

FORTHCOMING

- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No. 578 (February 2006).**
- M. BRATTI, D. CHECCHI and G. DE BLASIO, Does the expansion of higher education increase the equality of educational opportunities? Evidence from Italy, in R. Matoušek; D. Stavárek (eds.), Labour, TD No. 679 (June 2008).
- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, TD No. 683 (August 2008).
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, **TD No. 696 (December 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, **TD No. 698 (December 2008).**
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS-IBÁÑEZ, Bank risk and monetary policy, Journal of Financial Stability, TD No. 712 (May 2009).

- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, **TD No. 730** (October 2009).
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di Politica Economica, **TD No. 787 (January 2011).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, **TD No. 800 (March 2011).**
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, **TD No. 819 (September 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, **TD No. 835 (November 2011).**
- E. GENNARI and G. MESSINA, *How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data*, International Tax and Public Finance, **TD No. 844** (January 2012).
- S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics, **TD No. 879 (September 2012).**
- F. D'AMURI and G. PERI, *Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession,* Journal of the European Economic Association, **TD No. 886** (October 2012).